UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-Q

(Mark One)

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended March 31, 2023

or

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from ______ to _____

Commission file number 1-8729

UNISYS CORPORATION

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

38-0387840 (I.R.S. Employer

Identification No.)

801 Lakeview Drive, Suite 100 Blue Bell, Pennsylvania 19422

(215) 986-4011

(Address, zip code and telephone number, including area code, of principal executive offices)

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, par value \$.01	UIS	New York Stock Exchange

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (\$232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes \boxtimes No \square

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer	\boxtimes	Accelerated filer	
Non-accelerated filer		Smaller reporting company	
		Emerging growth company	

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes \Box No \boxtimes Number of shares of Unisys Common Stock, par value \$.01, outstanding as of March 31, 2023: 68,263,942

TABLE OF CONTENTS

PART I - F	INANCIAL INFORMATION	Page Number
Item 1.	Consolidated Financial Statements (Unaudited)	
	Consolidated Statements of Income (Loss)	<u>2</u>
	Consolidated Statements of Comprehensive Income (Loss)	<u>3</u>
	Consolidated Balance Sheets	<u>4</u>
	Consolidated Statements of Cash Flows	<u>5</u>
	Consolidated Statements of Equity (Deficit)	<u>6</u>
	Notes to Consolidated Financial Statements	Z
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>21</u>
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	<u>25</u>
Item 4.	Controls and Procedures	<u>25</u>
PART II -	OTHER INFORMATION	
Item 1.	Legal Proceedings	<u>27</u>
Item 1A.	Risk Factors	<u>27</u>
Item 6.	Exhibits	<u>28</u>
Exhibit Ind	ex	<u>29</u>
Signatures		<u>30</u>

Part I - FINANCIAL INFORMATION

Item 1. Financial Statements

UNISYS CORPORATION

CONSOLIDATED STATEMENTS OF INCOME (LOSS) (Unaudited)

(Millions, except per share data)

	Three Mo Marc	nths Er ch 31,	nded
	 2023	2	022
Revenue			
Services	\$ 403.9	\$	392.1
Technology	 112.5		54.6
	516.4		446.7
Costs and expenses			
Cost of revenue			
Services	316.1		321.3
Technology	 41.3		38.0
	357.4		359.3
Selling, general and administrative	102.9		104.4
Research and development	 6.2		6.5
	 466.5		470.2
Operating income (loss)	 49.9		(23.5)
Interest expense	7.6		8.4
Other (expense), net	(196.9)		(21.0)
Loss before income taxes	(154.6)		(52.9)
Provision for income taxes	19.9		4.1
Consolidated net loss	 (174.5)		(57.0)
Net income attributable to noncontrolling interests	0.9		0.3
Net loss attributable to Unisys Corporation	\$ (175.4)	\$	(57.3)
Loss per share attributable to Unisys Corporation	 		
Basic	\$ (2.58)	\$	(0.85)
Diluted	\$ (2.58)	\$	(0.85)

See notes to consolidated financial statements

CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (LOSS) (Unaudited)

(Millions)

	 Three Mor Marc	
	 2023	2022
Consolidated net loss	\$ (174.5)	\$ (57.0)
Other comprehensive income		
Foreign currency translation	24.4	(17.7)
Postretirement adjustments, net of tax of \$(3.0) in 2023 and \$6.7 in 2022	178.3	57.3
Total other comprehensive income	202.7	 39.6
Comprehensive income (loss)	 28.2	 (17.4)
Less comprehensive income (loss) attributable to noncontrolling interests	0.8	(0.9)
Comprehensive income (loss) attributable to Unisys Corporation	\$ 27.4	\$ (16.5)

See notes to consolidated financial statements

CONSOLIDATED BALANCE SHEETS (Unaudited)

(Millions)

~

~ 4

	Mar	rch 31, 2023	De	cember 31, 2022
Assets	14101	ch 51, 2025		2022
Current assets:				
Cash and cash equivalents	\$	391.9	\$	391.8
Accounts receivable, net		443.8	•	402.5
Contract assets		17.2		28.9
Inventories		15.9		14.9
Prepaid expenses and other current assets		108.9		92.3
Total current assets		977.7		930.4
Properties		397.9		410.8
Less-accumulated depreciation and amortization		324.6		334.9
Properties, net		73.3		75.9
Outsourcing assets, net		58.7		66.4
Marketable software, net		164.9		165.1
Operating lease right-of-use assets		39.0		42.5
Prepaid postretirement assets		115.5		119.5
Deferred income taxes		113.5		118.6
Goodwill		287.2		287.1
Intangible assets, net		49.9		52.4
Restricted cash		8.6		10.9
Assets held-for-sale		6.4		6.4
Other long-term assets		171.7		190.4
Total assets	\$	2,066.4	\$	2,065.6
Total liabilities and equity				
Current liabilities:				
Current maturities of long-term debt	\$	16.0	\$	17.4
Accounts payable		151.1		160.8
Deferred revenue		231.6		200.7
Other accrued liabilities		254.1		271.6
Total current liabilities		652.8		650.5
Long-term debt		490.1		495.7
Long-term postretirement liabilities		697.3		714.6
Long-term deferred revenue		115.3		122.3
Long-term operating lease liabilities		25.6		29.7
Other long-term liabilities		31.3		31.0
Commitments and contingencies (see Note 13)				
Equity:				
Common stock, shares issued: 2023; 73.9, 2022; 73.3		0.7		0.7
Accumulated deficit		(1,690.4)		(1,515.0)
Treasury stock, shares at cost: 2023; 5.6, 2022; 5.5		(156.3)		(156.0)
Paid-in capital		4,735.9		4,731.6
Accumulated other comprehensive loss		(2,873.2)		(3,076.0)
Total Unisys Corporation stockholders' equity (deficit)		16.7		(14.7)
Noncontrolling interests		37.3		36.5
Total equity		54.0		21.8
Total liabilities and equity	\$	2,066.4	\$	2,065.6

CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

(Millions)

		Three Months Ended March 31,				
		2023		2022		
Cash flows from operating activities						
Consolidated net loss	\$	(174.5)	\$	(57.0)		
Adjustments to reconcile consolidated net loss to net cash provided by (used for) operating activities:						
Foreign currency gains		(3.7)		(2.2)		
Non-cash interest expense		0.3		0.4		
Employee stock compensation		4.7		6.6		
Depreciation and amortization of properties		9.2		10.4		
Depreciation and amortization of outsourcing assets		12.2		18.3		
Amortization of marketable software		12.0		15.8		
Amortization of intangible assets		2.5		2.4		
Other non-cash operating activities		0.2		0.4		
Loss on disposal of capital assets		_		0.5		
Postretirement contributions		(16.4)		(16.2)		
Postretirement expense		193.2		10.2		
Deferred income taxes, net		6.3		(3.7)		
Changes in operating assets and liabilities, excluding the effect of acquisitions:						
Receivables, net and contract assets		(1.1)		94.2		
Inventories		(0.8)		(5.4)		
Other assets		(12.2)		(26.4)		
Accounts payable and current liabilities		(15.3)		(79.0)		
Other liabilities		(3.8)		(2.3)		
Net cash provided by (used for) operating activities		12.8		(33.0)		
Cash flows from investing activities						
Proceeds from investments		830.2		939.0		
Purchases of investments		(821.0)		(941.3)		
Investment in marketable software		(10.3)		(11.1)		
Capital additions of properties		(7.3)		(5.2)		
Capital additions of outsourcing assets		(2.7)		(2.4)		
Purchase of businesses, net of cash acquired		—		(0.3)		
Other		(0.4)		(0.4)		
Net cash used for investing activities		(11.5)		(21.7)		
Cash flows from financing activities						
Payments of long-term debt		(7.2)		(7.7)		
Other		(0.4)		(3.5)		
Net cash used for financing activities		(7.6)		(11.2)		
Effect of exchange rate changes on cash, cash equivalents and restricted cash		4.1		6.2		
Decrease in cash, cash equivalents and restricted cash		(2.2)	_	(59.7)		
Cash, cash equivalents and restricted cash, beginning of period		402.7		560.6		
Cash, cash equivalents and restricted cash, end of period	\$	400.5	\$	500.9		
	Ψ			000.0		

See notes to consolidated financial statements

CONSOLIDATED STATEMENTS OF EQUITY (DEFICIT) (Unaudited)

(Millions)

			Unisys Corporation												
	Tota	ıl	Unisys oration	Common Stock Par Value		Accumu- lated Deficit		Treasury Stock At Cost		Paid-in Capital		Accumu- lated Other Compre- hensive Loss			Non- controlling Interests
Balance at December 31, 2022	\$	21.8	\$ (14.7)	\$	0.7	\$	(1,515.0)	\$	(156.0)	\$	4,731.6	\$	(3,076.0)	\$	36.5
Consolidated net (loss) income	(1	.74.5)	(175.4)				(175.4)								0.9
Stock-based activity		4.0	4.0						(0.3)		4.3				
Translation adjustments		24.4	23.3										23.3		1.1
Postretirement plans	1	178.3	179.5										179.5		(1.2)
Balance at March 31, 2023	\$	54.0	\$ 16.7	\$	0.7	\$	(1,690.4)	\$	(156.3)	\$	4,735.9	\$	(2,873.2)	\$	37.3

						Unisys Cor	pora	tion						
	Total	Total Unisys Corporation Corporation			Treasury Accumu- Stock At lated Deficit Cost			Stock At	Paid-in Capital			Accumu- lated Other Compre- hensive Loss		Non- controlling Interests
Balance at December 31, 2021	\$ (64.4)	\$ (113.7)	\$	0.7	\$	(1,409.0)	\$	(152.2)	\$	4,710.9	\$	(3,264.1)	\$	49.3
Consolidated net (loss) income	(57.0)	(57.3)				(57.3)								0.3
Stock-based activity	2.2	2.2						(3.5)		5.7				
Translation adjustments	(17.7)	(14.9)										(14.9)		(2.8)
Postretirement plans	57.3	55.7										55.7		1.6
Balance at March 31, 2022	\$ (79.6)	\$ (128.0)	\$	0.7	\$	(1,466.3)	\$	(155.7)	\$	4,716.6	\$	(3,223.3)	\$	48.4

See notes to consolidated financial statements

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

(Dollars in millions, except share and per share amounts)

Note 1 - Basis of Presentation

The accompanying consolidated financial statements and footnotes of Unisys Corporation have been prepared in accordance with generally accepted accounting principles in the United States of America (GAAP). The financial statements and footnotes are unaudited. In the opinion of management, the financial information furnished herein reflects all adjustments necessary for a fair statement of the results of operations, comprehensive income (loss), financial position, cash flows and equity (deficit) for the interim periods specified. These adjustments consist only of normal recurring accruals except as disclosed herein. Because of seasonal and other factors, results for interim periods are not necessarily indicative of the results to be expected for the full year.

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions about future events. These estimates and assumptions affect the amounts of assets and liabilities reported, disclosures about contingent assets and liabilities and the reported amounts of revenue and expenses. Such estimates include the valuation of estimated credit losses, contract assets, operating lease right-of-use assets, outsourcing assets, marketable software, goodwill, purchased intangibles and other long-lived assets, legal contingencies, assumptions used in the calculation for systems integration projects, income taxes and retirement and other post-employment benefits, among others. These estimates and assumptions are based on management's best estimates and judgment. Management evaluates its estimates and assumptions on an ongoing basis using historical experience and other factors, including the current economic environment, which management believes to be reasonable under the circumstances. Management adjusts such estimates and assumptions when facts and circumstances dictate. As future events and their effects cannot be determined with precision, actual results could differ materially from these estimates. Changes in those estimates resulting from continuing changes in the economic environment such as rising interest rates, inflation, fluctuation in foreign exchange rates and the ongoing conflict in Ukraine, will be reflected in the financial statements in future periods.

The company's accounting policies are set forth in detail in Note 1 of the Notes to Consolidated Financial Statements in the company's Annual Report on Form 10-K for the year ended December 31, 2022 filed with the Securities and Exchange Commission. Such Annual Report also contains a discussion of the company's critical accounting policies and estimates. The company believes that these critical accounting policies and estimates affect its more significant estimates and judgments used in the preparation of the company's consolidated financial statements.

Note 2 - Cost-Reduction Actions

During the three months ended March 31, 2023, the company recognized net charges related to workforce reductions of \$0.7 million, principally related to severance costs. These net charges were comprised of: (a) a charge of \$2.6 million and (b) a credit of \$1.9 million for changes in estimates. In addition, the company recorded a credit of \$3.5 million for net foreign currency gains related to exiting foreign countries.

During the three months ended March 31, 2022, the company recognized net cost-reduction charges and other costs of \$3.0 million. The credit related to workforce reductions was \$0.6 million for changes in estimates. In addition, the company recorded net charges of \$3.6 million comprised of a charge of \$1.1 million for net foreign currency losses related to exiting foreign countries, a charge of \$3.8 million for asset impairments and a credit of \$1.3 million for changes in estimates related to other cost-reduction efforts.



The charges (credits) were recorded in the following statement of income (loss) classifications:

	Thre	ee Months 3	ed March
	2	2023	2022
Cost of revenue	\$	0.6	\$ 2.7
Selling, general and administrative		_	(0.7)
Research and development		0.1	(0.1)
Other (expense), net		(3.5)	1.1
Total	\$	(2.8)	\$ 3.0

Liabilities and expected future payments related to the company's work-force reduction actions are as follows:

	Т	Total		U.S.	Interna	ational
Balance at December 31, 2022	\$	11.7	\$	4.2	\$	7.5
Additional provisions		2.6		0.8		1.8
Payments		(2.8)		(1.4)		(1.4)
Changes in estimates		(1.9)		(0.7)		(1.2)
Translation adjustments		0.1		—		0.1
Balance at March 31, 2023	\$	9.7	\$	2.9	\$	6.8
Expected future utilization on balance at March 31, 2023:			_			
Short-term	\$	9.7	\$	2.9	\$	6.8

Note 3 - Pension and Postretirement Benefits

Net periodic pension expense (income) is presented below:

		 e Months End arch 31, 2023			Three Months Ended March 31, 2022							
	 Total	U.S. Plans	International Plans			Total		U.S. Plans	In	ternational Plans		
Service cost ⁽ⁱ⁾	\$ 0.3	\$ 	\$	0.3	\$	0.5	\$		\$	0.5		
Interest cost	58.5	40.8		17.7		38.9		28.3		10.6		
Expected return on plan assets	(69.4)	(48.6)		(20.8)		(68.4)		(47.5)		(20.9)		
Amortization of prior service benefit	(1.2)	(0.6)		(0.6)		(1.3)		(0.6)		(0.7)		
Recognized net actuarial loss	22.1	19.9		2.2		40.9		30.7		10.2		
Settlement losses ⁽ⁱⁱ⁾	183.2	183.2				_				—		
Net periodic pension expense (income)	\$ 193.5	\$ 194.7	\$	(1.2)	\$	10.6	\$	10.9	\$	(0.3)		

⁽ⁱ⁾Service cost is reported in selling, general and administrative expense. All other components of net periodic pension expense (income) are reported in other (expense), net in the consolidated statements of income (loss).

⁽ⁱⁱ⁾In March 2023, the company purchased a group annuity contract, with plan assets, for approximately \$265 million to transfer projected benefit obligations related to approximately 8,650 retirees of one of the company's U.S. defined benefit pension plans. This action resulted in a pre-tax settlement loss of \$183.2 million for the three months ended March 31, 2023.

In 2023, the company expects to make cash contributions of approximately \$40 million primarily for its international defined benefit pension plans. In 2022, the company made cash contributions of \$39.3 million to its worldwide defined benefit pension plans. During the three months ended March 31, 2023 and 2022, the company made cash contributions of \$14.5 million and \$15.1 million, respectively.

At the end of each year, the company estimates its future cash contributions to its U.S. qualified defined benefit pension plans based on year-end pension data and assumptions. Any material deterioration in the value of the company's U.S. qualified defined benefit pension plan assets, as well as changes in pension legislation, discount rate changes, asset return changes, or changes in economic or demographic trends, could require the company to make cash contributions to its U.S. defined benefit pension plans in different amounts and on a different schedule than previously contemplated.

Net periodic postretirement benefit income is presented below:

		Ended ,		
		2023		2022
Service cost ⁽ⁱ⁾	\$	_	\$	—
Interest cost		0.8		0.5
Expected return on assets		(0.1)		(0.1)
Recognized net actuarial gain		(0.7)		(0.5)
Amortization of prior service benefit		(0.3)		(0.3)
Net periodic postretirement benefit income	\$	(0.3)	\$	(0.4)

⁽ⁱ⁾Service cost is reported in selling, general and administrative expense. All other components of net periodic postretirement benefit expense (income) are reported in other (expense), net in the consolidated statements of income (loss).

The company expects to make cash contributions of \$4.0 million to its postretirement benefit plan in 2023. In 2022, the company made cash contributions of \$4.3 million to its postretirement benefit plan. For the three months ended March 31, 2023 and 2022, the company made cash contributions of \$1.9 million and \$1.1 million, respectively.

Note 4 - Stock Compensation

Under stockholder approved stock-based plans, stock options, stock appreciation rights, restricted stock and restricted stock units may be granted to officers, directors and other key employees.

As of March 31, 2023, the company has granted restricted stock and restricted stock units under these plans. The company recognizes compensation cost, net of a forfeiture rate, in selling, general and administrative expense, and recognizes compensation cost only for those awards expected to vest. The company estimates the forfeiture rate based on its historical experience and its expectations about future forfeitures.



During the three months ended March 31, 2023 and 2022, the company recorded \$4.7 million and \$6.6 million of share-based restricted stock and restricted stock unit compensation expense, respectively.

Restricted stock and restricted stock unit awards may contain time-based units, performance-based units, total shareholder return market-based units, or a combination of these units. Each performance-based and market-based unit will vest into zero to two shares depending on the degree to which the performance or market conditions are met. Compensation expense for performance-based awards is recognized as expense ratably for each installment from the date of grant until the date the restrictions lapse and is based on the fair market value at the date of grant and the probability of achievement of the specific performance-related goals. Compensation expense for market-related awards is recognized as expense ratably over the measurement period, regardless of the actual level of achievement, provided the service requirement is met. Restricted stock unit grants for the company's directors vest upon award and compensation expense for such awards is recognized upon grant.

A summary of restricted stock and restricted stock unit (RSU) activity for the three months ended March 31, 2023 follows (shares in thousands):

	Restricted Stock and RSU	A Gra	eighted- verage ant-Date ir Value
Outstanding at December 31, 2022	2,230	\$	23.53
Granted	2,314		5.42
Vested	(529)		12.39
Forfeited and expired	(206)		23.39
Outstanding at March 31, 2023	3,809		12.30

The aggregate weighted-average grant-date fair value of restricted stock and restricted stock units granted during the three months ended March 31, 2023 and 2022 was \$13.8 million and \$22.2 million, respectively. The fair value of restricted stock and restricted stock units with time and performance conditions was determined based on the trading price of the company's common shares on the date of grant. The fair value of awards with market conditions was estimated using a Monte Carlo simulation with the following weighted-average assumptions:

	_	Three Months Ended March 31,				
		2023		2022		
Weighted-average fair value of grant	\$	7.32	\$	34.14		
Risk-free interest rate ⁽ⁱ⁾		4.51 %		1.72 %		
Expected volatility ⁽ⁱⁱ⁾		63.63 %	57.71 %			
Expected life of restricted stock units in years ⁽ⁱⁱⁱ⁾		2.84		2.85		
Expected dividend yield		— %		— %		

⁽ⁱ⁾Represents the continuously compounded semi-annual zero-coupon U.S. treasury rate commensurate with the remaining performance period. ⁽ⁱⁱ⁾Based on historical volatility for the company that is commensurate with the length of the performance period. ⁽ⁱⁱⁱ⁾Represents the remaining life of the longest performance period.

As of March 31, 2023, there was \$30.8 million of total unrecognized compensation cost related to outstanding restricted stock and restricted stock units granted under the company's plans. That cost is expected to be recognized over a weighted-average period of 2.2 years. The aggregate weighted-average grant-date fair value of restricted stock and restricted stock units vested during the three months ended March 31, 2023 and 2022 was \$6.6 million and \$14.0 million, respectively.

Common stock issued upon the lapse of restrictions on restricted stock and restricted stock units are newly issued shares. In light of its tax position, the company is currently not recognizing any tax benefits from the issuance of stock upon lapse of restrictions on restricted stock and restricted stock units.

Note 5 - Other (expense), net

Other (expense), net is comprised of the following:

	Three Months Ended March 31,				
	 2023		2022		
Postretirement expense*	\$ (192.9)	\$	(9.7)		
Foreign exchange gains**	3.7		2.2		
Environmental costs and other, net***	(7.7)		(13.5)		
Total other (expense), net	\$ (196.9)	\$	(21.0)		

*Includes \$183.2 million in the three months ended March 31, 2023 of a settlement loss related to one of the company's U.S. defined benefit pension plans. See Note 3.

**Includes net foreign exchange gains of \$3.5 million and net foreign exchange losses of \$1.1 million, respectively, in the three months ended March 31, 2023 and 2022, related to substantial completion of liquidation of foreign subsidiaries.

***Environmental costs relate to a previously disposed business.

Note 6 - Income Taxes

Accounting rules governing income taxes require that deferred tax assets and liabilities be recognized using enacted tax rates for the effect of temporary differences between the book and tax bases of recorded assets and liabilities. These rules also require that deferred tax assets be reduced by a valuation allowance if it is more likely than not that some portion or the entire deferred tax asset will not be realized.

The company evaluates the realizability of its deferred tax assets by assessing its valuation allowance and by adjusting such amount, if necessary. The realization of the company's net deferred tax assets as of March 31, 2023 is primarily dependent on the ability to generate sustained taxable income in various jurisdictions. Judgment is required to estimate forecasted future taxable income, which may be impacted by future business developments, actual results, strategic operational and tax initiatives, legislative, and other economic factors and developments. Any increase or decrease in the valuation allowance would result in additional or lower income tax expense in that period and could have a significant impact on that period's earnings.

A full valuation allowance is currently maintained for all U.S. and certain foreign deferred tax assets in excess of deferred tax liabilities. The company will record a tax provision or benefit for those international subsidiaries that do not have a full valuation allowance against their net deferred tax assets. Any profit or loss recorded for the company's U.S. operations will have no provision or benefit associated with it due to such valuation allowance, except with respect to withholding taxes not creditable against future taxable income. As a result, the company's provision or benefit for taxes may vary significantly depending on the geographic distribution of income.

A corporation's ability to deduct its federal net operating loss (NOL) carryforwards and utilize certain other available tax attributes can be substantially constrained under the general annual limitation rules of Section 382 of the U.S. Internal Revenue Code (Section 382) if it undergoes an "ownership change" as defined in Section 382 (generally where cumulative stock ownership changes among material shareholders exceed 50 percent during a rolling three-year period). Similar rules may apply under state tax laws. A future tax "ownership change" pursuant to Section 382 or future changes in tax laws that impose tax attribute utilization limitations may severely limit or effectively eliminate the company's ability to utilize its NOL carryforwards and other tax attributes.

Note 7 - Loss Per Share

The following table shows how loss per share attributable to Unisys Corporation was computed (shares in thousands):

		nded		
		2023	2	2022
Basic loss per common share computation:				
Net loss attributable to Unisys Corporation	\$	(175.4)	\$	(57.3)
Weighted average shares		67,943		67,387
Basic loss per common share	\$	(2.58)	\$	(0.85)
Diluted loss per common share computation:				
Net loss attributable to Unisys Corporation	\$	(175.4)	\$	(57.3)
Weighted average shares		67,943		67,387
Plus incremental shares from assumed conversions of employee stock plans				
Adjusted weighted average shares	_	67,943		67,387
Diluted loss per common share	\$	(2.58)	\$	(0.85)
Anti-dilutive weighted-average stock options and restricted stock units ⁽ⁱ⁾		391		837

⁽ⁱ⁾Amounts represent shares excluded from the computation of diluted loss per share, as their effect, if included, would have been anti-dilutive for the periods presented.

Note 8 - Contract Assets and Deferred Revenue

Contract assets represent rights to consideration in exchange for goods or services transferred to a customer when that right is conditional on something other than the passage of time. Deferred revenue represents contract liabilities.

Net contract assets (liabilities) are as follows:

	March	31, 2023	De	ecember 31, 2022
Contract assets - current	\$	17.2	\$	28.9
Contract assets - long-term ⁽ⁱ⁾		10.6		11.0
Deferred revenue - current		(231.6)		(200.7)
Deferred revenue - long-term		(115.3)		(122.3)

⁽ⁱ⁾Reported in other long-term assets on the company's consolidated balance sheets.

Significant changes in the above contract liability balances were as follows:

]	Three Months Endee March 31,			
	2023			2022	
Revenue recognized that was included in deferred revenue at the beginning of the period	\$	65.1	\$	82.9	

Note 9 - Capitalized Contract Costs

The company's incremental direct costs of obtaining a contract consist of sales commissions which are deferred and amortized ratably over the initial contract life. These costs are classified as current or noncurrent based on the timing of when the company expects to recognize the expense. The current and noncurrent portions of deferred commissions are included in prepaid expenses, other current assets and in other long-term assets, respectively, in the company's consolidated balance sheets. At March 31, 2023 and December 31, 2022, the company had \$3.2 million and \$4.9 million, respectively, of deferred commissions.

Amortization expense related to deferred commissions was as follows:

	Three Months Ended March 31,				
				2022	
Deferred commissions - amortization expense ⁽ⁱ⁾	\$	0.4	\$	1.1	

⁽ⁱ⁾Reported in selling, general and administrative expense in the company's consolidated statements of income (loss).

Costs on outsourcing contracts are generally expensed as incurred. However, certain costs incurred upon initiation of an outsourcing contract (costs to fulfill a contract), principally initial customer setup, are capitalized and expensed over the initial contract life. These costs are included in outsourcing assets, net in the company's consolidated balance sheets. The amount of such costs at March 31, 2023 and December 31, 2022 was \$31.4 million and \$34.8 million, respectively. These costs are amortized over the initial contract life and reported in cost of revenue.

Amortization expense related to costs to fulfill a contract was as follows:

	Three Mor Marc	ıded	
	2023	2	2022
Costs to fulfill a contract - amortization expense	\$ 2.2	\$	8.9

The remaining balance of outsourcing assets, net is comprised of fixed assets and software used in connection with outsourcing contracts. These costs are capitalized and depreciated over the shorter of the initial contract life or in accordance with the company's fixed asset policy.

Note 10 - Financial Instruments and Fair Value Measurements

Due to its foreign operations, the company is exposed to the effects of foreign currency exchange rate fluctuations on the U.S. dollar, principally related to intercompany account balances. The company uses derivative financial instruments to reduce its exposure to market risks from changes in foreign currency exchange rates on such balances. The company enters into foreign exchange forward contracts, generally having maturities of three months or less, which have not been designated as hedging instruments. At March 31, 2023 and December 31, 2022, the notional amount of these contracts was \$468.7 million and \$533.5 million, respectively. The fair value of these forward contracts is based on quoted prices for similar but not identical financial instruments; as such, the inputs are considered Level 2 inputs.

The following table summarizes the fair value of the company's foreign exchange forward contracts.

	March 31,	2023	ember 31, 2022
Balance Sheet Location			
Prepaid expenses and other current assets	\$	3.1	\$ 7.9
Other accrued liabilities		0.5	1.3
Total fair value	\$	2.6	\$ 6.6

The following table summarizes the location and amount of gains and (losses) recognized on foreign exchange forward contracts.

	Th	Ended		
	2023			2022
Statement of Income Location				
Other (expense), net	\$	5.2	\$	(10.9)

Financial assets with carrying values approximating fair value include cash and cash equivalents and accounts receivable. Financial liabilities with carrying values approximating fair value include accounts payable and other liabilities. The carrying amounts of these financial assets and liabilities approximate fair value due to their short maturities. Such financial instruments are not included in the following table that provides information about the estimated fair values of other financial instruments that are not measured at fair value in the consolidated balance sheets as of March 31, 2023 and December 31, 2022.

	March 31, 2023				2022		
	Carrying Amount	Fai	ir Value		Carrying Amount	Fai	ir Value
Long-term debt:							
6.875% senior secured notes due November 1, 2027	\$ 479.5	\$	309.8	\$	479.2	\$	373.0

Long-term debt is carried at amortized cost and its estimated fair value is based on market prices classified as Level 2 in the fair value hierarchy.

Note 11 - Goodwill and Intangible Assets

Goodwill

Changes in the carrying value of goodwill by reporting unit were as follows:

	Total	DWS	CA&I	ECS	Other
Balance at December 31, 2022	\$ 287.1	\$ 140.5	\$ 38.0	\$ 98.3	\$ 10.3
Translation adjustments	0.1	0.1	—	—	—
Balance at March 31, 2023	\$ 287.2	\$ 140.6	\$ 38.0	\$ 98.3	\$ 10.3

At March 31, 2023, there was no goodwill allocated to reporting units with negative net assets.

Intangible Assets, Net

Intangible assets, net at March 31, 2023 consists of the following:

	Gross Carry	ring Amount	Accumulated Amortization	N	Vet Carrying Amount
Technology ⁽ⁱ⁾	\$	10.0	\$ 5.7	\$	4.3
Customer relationships ⁽ⁱⁱ⁾		54.2	9.5		44.7
Marketing ⁽ⁱⁱ⁾		1.3	0.4		0.9
Total	\$	65.5	\$ 15.6	\$	49.9

⁽ⁱ⁾ Amortization expense is included within cost of revenue - technology in the consolidated statements of income (loss).

⁽ⁱⁱ⁾ Amortization expense is included within selling, general and administrative expense in the consolidated statements of income (loss).

For the three months ended March 31, 2023 and 2022 amortization expense was \$2.5 million and \$2.4 million, respectively.

The future amortization relating to acquired intangible assets at March 31, 2023 was estimated as follows:

	Future Amortizat Expense					
Remainder of 2023	\$	7.1				
2024		7.2				
2025		4.3				
2026		4.0				
2027		4.0				
Thereafter		23.3				
Total	\$	49.9				

Note 12 - Debt

Long-term debt is comprised of the following:

	March 31, 2023		Dece	mber 31, 2022
6.875% senior secured notes due November 1, 2027 (Face value of \$485.0 million less unamortized issuance costs of \$5.5 and \$5.8 million at March 31, 2023 and at December 31, 2022)	\$	479.5	\$	479.2
Finance leases		0.8		1.1
Other debt		25.8		32.8
Total		506.1		513.1
Less – current maturities		16.0		17.4
Total long-term debt	\$	490.1	\$	495.7

See Note 10 for the fair value of the notes.

Senior Secured Notes due 2027

The company has outstanding \$485.0 million aggregate principal amount of its 6.875% Senior Secured Notes due 2027 (the 2027 Notes). The 2027 Notes pay interest semiannually on May 1 and November 1 and will mature on November 1, 2027, unless earlier repurchased or redeemed. The 2027 Notes are fully and unconditionally guaranteed on a senior secured basis by Unisys Holding Corporation, Unisys NPL, Inc. and Unisys AP Investment Company I, each of which is a U.S. corporation that is directly or indirectly owned by the company (the subsidiary guarantors).

The 2027 Notes and the related guarantees rank equally in right of payment with all of the existing and future senior debt of the company and its subsidiary guarantors and senior in right of payment to any future subordinated debt of the company and its subsidiary guarantors. The 2027 Notes and the related guarantees are structurally subordinated to all existing and future liabilities (including preferred stock, trade payables and pension liabilities) of the subsidiaries of the company that are not subsidiary guarantors. The 2027 Notes and the guarantees are secured by liens on substantially all assets of the company and the subsidiary guarantors, other than certain excluded assets (the collateral). The liens securing the 2027 Notes on certain ABL collateral are subordinated to the liens on ABL collateral in favor of the ABL secured parties and, in the future, the liens securing the 2027 Notes may be subordinated to liens on the collateral securing certain permitted first lien debt, subject to certain limitations and permitted liens.

Prior to November 1, 2023, the company may, at its option, redeem some or all of the 2027 Notes at any time, at a price equal to 100% of the principal amount of the 2027 Notes redeemed plus a "make-whole" premium, plus accrued and unpaid interest, if any. The company may also redeem, at its option, up to 40% of the 2027 Notes at any time prior to November 1, 2023, using the proceeds of certain equity offerings at a redemption price of 106.875% of the principal amount thereof, plus accrued and unpaid interest, if any. On or after November 1, 2023, the company may, on any one or more occasions, redeem all or a part of the 2027 Notes at specified redemption premiums, declining to par for any redemptions on or after November 1, 2025.

The indenture contains covenants that limit the ability of the company and its restricted subsidiaries to, among other things: (i) incur additional indebtedness and guarantee indebtedness; (ii) pay dividends or make other distributions or repurchase or redeem its capital stock; (iii) prepay, redeem or repurchase certain debt; (iv) make certain prepayments in respect of pension obligations; (v) issue certain preferred stock or similar equity securities; (vi) make loans and investments (including investments by the company and subsidiary guarantors in subsidiaries that are not guarantors); (vii) sell assets; (viii) create or incur liens; (ix) enter into transactions with affiliates; (x) enter into agreements restricting its subsidiaries' ability to pay dividends; and (xi) consolidate, merge or sell all or substantially all of its assets. These covenants are subject to several important limitations and exceptions.

If the company experiences certain kinds of changes of control (as defined in the indenture), it will be required to offer to repurchase the 2027 Notes at 101% of the principal amount of the 2027 Notes, plus accrued and unpaid interest as of the repurchase date, if any. In addition, if the company sells assets, under certain circumstances it must apply the proceeds towards an offer to repurchase the 2027 Notes at a price equal to par plus accrued and unpaid interest, if any.

The indenture also provides for events of default, which, if any of them occur, would permit or require the principal, premium, if any, interest and any other monetary obligations on all the then outstanding 2027 Notes to be due and payable immediately.

Interest expense related to the 2027 Notes is comprised of the following:

		Three Months Ende March 31,		
	2	2023	2	2022
Contractual interest coupon	\$	8.3	\$	8.3
Amortization of issuance costs		0.3		0.3
Total	\$	8.6	\$	8.6

Other Debt

The company has a \$27.7 million Installment Payment Agreement (IPA) maturing on December 20, 2023 with a syndicate of financial institutions to finance the acquisition of certain software licenses necessary for the provision of services to a client. Interest accrues at an annual rate of 7.0% and the company is required to make monthly principal and interest payments on each agreement in arrears. At March 31, 2023 and December 31, 2022, \$4.1 million and \$5.5 million, was reported in current maturities of long-term debt, respectively.

The company has a vendor agreement in the amount of \$19.3 million to finance the acquisition of certain software licenses used to provide services to our clients and for its own internal use. Interest accrues at an annual rate of 5.47% and the company is required to make annual principal and interest payments in advance with the last payment due on March 1, 2024. At March 31, 2023 and December 31, 2022, \$4.2 million and \$4.0 million, was reported in current maturities of long-term debt, respectively.

Asset Based Lending (ABL) Credit Facility

The company has a secured revolving credit facility (the Amended and Restated ABL Credit Facility) that matures on October 29, 2025 and provides for revolving loans and letters of credit up to an aggregate amount of \$145.0 million (with a limit on letters of credit of \$40.0 million), with an accordion feature provision allowing for the aggregate amount available under the credit facility to be increased up to \$175.0 million upon the satisfaction of certain conditions specified in the Amended and Restated ABL Credit Facility. Availability under the credit facility is subject to a borrowing base calculated by reference to the company's receivables. At March 31, 2023, the company had no borrowings and \$6.6 million of letters of credit outstanding, and availability under the facility was \$64.0 million net of letters of credit issued.

The Amended and Restated ABL Credit Facility is subject to a springing maturity, under which the Amended and Restated ABL Credit Facility will immediately mature 91 days prior to any date on which contributions to pension funds in the United States in an amount in excess of \$100.0 million are required to be paid unless the company is able to meet certain conditions, including that the company has the liquidity (as defined in the Amended and Restated ABL Credit Facility) to cash settle the amount of such pension payments, no default or event of default has occurred under the Amended and Restated ABL Credit Facility, the company's liquidity is above \$130.0 million and the company is in compliance with the then applicable fixed charge coverage ratio on a pro forma basis.

The Amended and Restated ABL Credit Facility is guaranteed by the subsidiary guarantors and any future material domestic subsidiaries. The facility is secured by the assets of the company and the subsidiary guarantors, other than certain excluded assets, under a security agreement entered into by the company and the subsidiary guarantors in favor of JPMorgan Chase Bank, N.A., as agent for the lenders under the credit facility.

The company is required to maintain a minimum fixed charge coverage ratio if the availability under the Amended and Restated ABL Credit Facility falls below the greater of 10% of the lenders' commitments under the facility and \$14.5 million.



The Amended and Restated ABL Credit Facility contains customary representations and warranties, including, but not limited to, that there has been no material adverse change in the company's business, properties, operations or financial condition. The Amended and Restated ABL Credit Facility includes restrictions on the ability of the company and its subsidiaries to, among other things, incur other debt or liens, dispose of assets and make acquisitions, loans and investments, repurchase its equity, and prepay other debt. These restrictions are subject to several important limitations and exceptions. Events of default include non-payment, failure to comply with covenants, materially incorrect representations and warranties, change of control and default under other debt aggregating at least \$50.0 million, subject to relevant cure periods, as applicable.

At March 31, 2023, the company has met all covenants and conditions under its various lending and funding agreements. For at least the next 12 months, the company expects to continue to meet these covenants and conditions.

Note 13 - Litigation and Contingencies

The company is involved in a wide range of lawsuits, claims, investigations and proceedings, which arise in the ordinary course of business, including actions with respect to commercial and government contracts, labor and employment, employee benefits, environmental matters, intellectual property and non-income tax matters. Further, given the rapidly evolving external landscape of cybersecurity, privacy and data protection laws, regulations and threat actors, the company and its clients have been and will continue to be subject to actions or proceedings in various jurisdictions. These matters can involve a number of different parties, including competitors, clients, current or former employees, government and regulatory agencies, stockholders and representatives of the locations in which the company does business.

The company records a provision for these matters when it is both probable that a liability has been incurred and the amount of the loss can be reasonably estimated. Significant judgment is required in both the determination of probability and the determination as to whether an exposure is reasonably estimable. Because of uncertainties related to these matters, accruals are based only on the best information available at the time. Any provisions are reviewed at least quarterly and are adjusted to reflect the impact and status of settlements, rulings, advice of counsel and other information and events pertinent to a particular matter. These adjustments could have a material impact on our results of operations and financial position.

The company intends to defend itself vigorously with respect to legal matters pending against it. Based on its experience, the company also believes that the damage amounts claimed in the matters disclosed below are not a meaningful indicator of the company's potential liability.

Litigation is inherently unpredictable and unfavorable resolutions could occur. Whether any losses, damages or remedies finally determined in any claim, suit, investigation or proceeding could reasonably have a material effect on the company's business, financial condition, results of operations or cash flows will depend on a number of variables, including: the timing and amount of such losses or damages; the structure and type of any such remedies; the significance of the impact any such losses, damages or remedies may have in the company's consolidated financial statements; and the unique facts and circumstances of the particular matter that may give rise to additional factors. Accordingly, it is possible that an adverse outcome from such matters could be material to the company's financial condition, results of operations and cash flows in any particular reporting period.

Notwithstanding that the ultimate results of the lawsuits, claims, investigations and proceedings that have been brought or asserted against the company are not currently determinable, the company believes that at March 31, 2023, it has adequate provisions for any such matters.

The following is a summary of the more significant legal matters involving the company.

The company's Brazilian operations, along with those of many other companies doing business in Brazil, are involved in various litigation matters, including numerous governmental assessments related to indirect and other taxes, as well as disputes associated with former employees and contract labor. The tax-related matters pertain to value-added taxes, customs, duties, sales and other non-income-related tax exposures. The labor-related matters include claims related to compensation. The company believes that appropriate accruals have been established for such matters based on information currently available. At March 31, 2023, excluding those matters that have been assessed by management as being remote as to the likelihood of ultimately resulting in a loss, the amount related to unreserved tax-related matters, inclusive of any related interest, is estimated to be up to approximately \$111 million.

On November 11, 2022, a purported stockholder of the company filed a putative securities class action complaint in the United States District Court for the Eastern District of Pennsylvania against the company and certain of its current officers, alleging violations of the Securities Exchange Act of 1934, as amended, based on allegedly false or misleading statements related to projections and certain other statements positively characterizing the company's momentum, business, prospects and operations, and the effectiveness of the company's internal control over financial reporting and the company's disclosure controls and procedures. The plaintiff seeks an award of compensatory damages, among other relief, and costs and attorneys' and experts' fees.

With respect to the specific legal proceedings and claims described above, except as otherwise noted, either (i) the amount or range of possible losses in excess of amounts accrued, if any, is not reasonably estimable or (ii) the company believes that the amount or range of possible losses in excess of amounts accrued that are estimable would not be material.

Note 14 - Accumulated Other Comprehensive Loss

Accumulated other comprehensive loss is as follows:

	Total	Translation Adjustments	Postretirement Plans
Balance at December 31, 2022	\$ (3,076.0)	\$ (977.4)	\$ (2,098.6)
Other comprehensive income (loss) before reclassifications	3.3	26.8	(23.5)
Amounts reclassified from accumulated other comprehensive loss	199.5	(3.5)	203.0
Current period other comprehensive income	202.8	23.3	 179.5
Balance at March 31, 2023	\$ (2,873.2)	\$ (954.1)	\$ (1,919.1)

Amounts reclassified out of accumulated other comprehensive loss are as follows:

	Three Moi Marc	
	 2023	2022
Translation adjustments:		
Adjustment for substantial completion of liquidation of foreign subsidiaries ⁽ⁱ⁾	\$ (3.5)	\$ 1.1
Postretirement plans ⁽ⁱⁱ⁾ :		
Amortization of prior service benefit	(1.3)	(1.7)
Amortization of actuarial losses	21.5	40.4
Settlement losses	183.2	_
Total before tax	199.9	 39.8
Income tax	(0.4)	(1.7)
Total reclassifications for the period	\$ 199.5	\$ 38.1

(i) Reported in other (expense), net in the consolidated statements of income (loss).
 (ii) These items are included in net periodic postretirement cost (see Note 3).

Note 15 - Supplemental Cash Flow Information

	 Three Months Ended March 31,		
	 2023		2022
Cash paid during the period for:		-	
Income taxes, net of refunds	\$ 20.7	\$	18.9
Interest	\$ 1.0	\$	1.4

The following table provides a reconciliation of cash and cash equivalents and restricted cash reported within the consolidated balance sheets to the total of the amounts shown in the consolidated statements of cash flows.

	Marc	ch 31, 2023	D	ecember 31, 2022
Cash and cash equivalents	\$	391.9	\$	391.8
Restricted cash		8.6		10.9
Total cash, cash equivalents and restricted cash shown in the consolidated statements of cash flows	\$	400.5	\$	402.7

Cash and cash equivalents subject to contractual restrictions and are therefore not readily available are classified as restricted cash.

Note 16 - Segment Information

The company's reportable segments are as follows:

- Digital Workplace Solutions (DWS), which provides modern and traditional workplace solutions;
- Cloud, Applications & Infrastructure Solutions (CA&I), which provides digital platform, applications, and infrastructure solutions; and
- Enterprise Computing Solutions (ECS), which provides solutions that harness secure, continuous high-intensity computing and enable digital services through software-defined operating environments.

The accounting policies of each segment are the same as those followed by the company as a whole. The company evaluates segment performance based on gross profit exclusive of the service cost component of postretirement income or expense, restructuring charges, amortization of purchased intangibles and unusual and nonrecurring items, which are included in other gross profit.

A summary of the company's operations by segment is presented below:

	Total Segments DWS		CA&I			ECS	
Three Months Ended March 31, 2023							
Revenue	\$	445.2	\$ 131.0	\$	126.0	\$	188.2
Gross profit	\$	157.5	\$ 15.6	\$	16.4	\$	125.5
Three Months Ended March 31, 2022							
Revenue	\$	374.5	\$ 124.8	\$	129.1	\$	120.6
Gross profit	\$	85.8	\$ 16.0	\$	7.0	\$	62.8

Presented below is a reconciliation of total segment revenue to total consolidated revenue:

	Three Mo Mare		
	2023	2022	
Total segment revenue	\$ 445.2	\$ 374.5	
Other revenue	71.2	72.2	
Total consolidated revenue	\$ 516.4	\$ 446.7	

Presented below is a reconciliation of total segment gross profit to consolidated loss before income taxes:

	 Three Moi Marc		
	 2023		2022
Total segment gross profit	\$ 157.5	\$	85.8
Other gross profit	1.5		1.6
Total gross profit	 159.0		87.4
Selling, general and administrative expense	(102.9)		(104.4)
Research and development expense	(6.2)		(6.5)
Interest expense	(7.6)		(8.4)
Other (expense), net	 (196.9)		(21.0)
Total loss before income taxes	\$ (154.6)	\$	(52.9)

Other revenue and other gross profit are comprised of an aggregation of a number of immaterial business activities and cost reductions charges. These businesses principally provide for the management of processes and functions for clients in select industries, helping them improve performance and reduce costs.

Geographic information about the company's revenue, which is principally based on location of the selling organization, is presented below:

	Three Months Ended March 31,			
	 2023		2022	
United States	\$ 201.0	\$	199.0	
United Kingdom	121.9		52.9	
Other foreign	193.5		194.8	
Total	\$ 516.4	\$	446.7	

Note 17 - Remaining Performance Obligations

Remaining performance obligations represent the transaction price of firm orders for which work has not been performed and excludes (1) contracts with an original expected length of one year or less and (2) contracts for which the company recognizes revenue at the amount to which it has the right to invoice for services performed. At March 31, 2023, the company had approximately \$0.6 billion of remaining performance obligations of which approximately 26% is estimated to be recognized as revenue by the end of 2023, 28% by the end of 2024, 19% by the end of 2025, 15% by the end of 2026 and 12% thereafter.

2	1	٦
2	l	J

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

This discussion and analysis of the company's financial condition and results of operations should be read in conjunction with the consolidated financial statements and the related notes included elsewhere in this quarterly report. In this discussion and analysis of the company's financial condition and results of operations, the company has included information that may constitute "forward-looking" statements, as defined in the Private Securities Litigation Reform Act of 1995. Forward-looking statements provide current expectations of future events and include any statement that does not directly relate to any historical or current fact. Words such as "anticipates," "believes," "expects," "intends," "plans," "projects" and similar expressions may identify such forward-looking statements. All forward-looking statements rely on assumptions and are subject to risks, uncertainties and other factors that could cause the company's actual results to differ materially from expectations. Factors that could affect future results include, but are not limited to, those discussed under "Risk Factors" in Part II, Item 1A. Any forward-looking statement speaks only as of the date on which that statement is made.

Overview

In March 2023, the company purchased a group annuity contract, with plan assets, for approximately \$265 million to transfer projected benefit obligations related to approximately 8,650 retirees of one of the company's U.S. defined benefit pension plans resulting in a pre-tax settlement loss of \$183.2 million for the three months ended March 31, 2023.

For the three months ended March 31, 2023, the company reported net loss attributable to Unisys Corporation of \$175.4 million, or \$2.58 per diluted share, compared with a loss of \$57.3 million, or \$0.85 per diluted share, for the three months ended March 31, 2022. Included in the loss for the three months ended March 31, 2023 was the U.S. pension settlement loss of \$183.2 million described above.

Results of operations

Company results

Three months ended March 31, 2023 compared with the three months ended March 31, 2022

Revenue for the three months ended March 31, 2023 was \$516.4 million compared with \$446.7 million for the three months ended March 31, 2022, an increase of 15.6% from the prior year. The increase was primarily due to higher software license renewals within the Enterprise Computing Solutions segment. Foreign currency fluctuations had a 3 percentage-point negative impact on revenue in the current period compared with the year-ago period.

U.S. revenue increased 1.0% in the current period compared with the year-ago period. International revenue increased 27.3% in the current period compared with the prior-year period, principally due to increases in Europe and Latin America. Foreign currency had a 7 percentage-point negative impact on international revenue in the three months ended March 31, 2023 compared with the three months ended March 31, 2022.

During the three months ended March 31, 2023, the company recognized net charges related to workforce reductions of \$0.7 million, principally related to severance costs. These net charges were comprised of: (a) a charge of \$2.6 million and (b) a credit of \$1.9 million for changes in estimates. In addition, the company recorded a credit of \$3.5 million for net foreign currency gains related to exiting foreign countries.

During the three months ended March 31, 2022, the company recognized net cost-reduction charges and other costs of \$3.0 million. The credit related to workforce reductions was \$0.6 million for changes in estimates. In addition, the company recorded net charges of \$3.6 million comprised of a charge of \$1.1 million for net foreign currency losses related to exiting foreign countries, a charge of \$3.8 million for asset impairments and a credit of \$1.3 million for changes in estimates related to other cost-reduction efforts.



The charges (credits) were recorded in the following statement of income (loss) classifications:

	Three Months Ended March 31,			
	2	023		2022
Cost of revenue	\$	0.6	\$	2.7
Selling, general and administrative		_		(0.7)
Research and development		0.1		(0.1)
Other (expense), net		(3.5)		1.1
Total	\$	(2.8)	\$	3.0

Gross profit and gross profit margin were \$159.0 million and 30.8% in the three months ended March 31, 2023, respectively, compared with \$87.4 million and 19.6% for the three months ended March 31, 2022, respectively. The increase was principally due to higher software license renewals.

Selling, general and administrative expense in the three months ended March 31, 2023 was \$102.9 million (19.9% of revenue) compared with \$104.4 million (23.4% of revenue) for the three months ended March 31, 2022.

Research and development (R&D) expense for the three months ended March 31, 2023 and 2022 was \$6.2 million and \$6.5 million, respectively.

For the three months ended March 31, 2023, the company reported an operating profit of \$49.9 million compared with an operating loss of \$23.5 million in the three months ended March 31, 2022. The increase was primarily driven by higher revenue and gross profit.

Interest expense for the three months ended March 31, 2023 and 2022 was \$7.6 million and \$8.4 million, respectively.

Other (expense), net was expense of \$196.9 million for the three months ended March 31, 2023 compared with expense of \$21.0 million for the three months ended March 31, 2022. Other (expense), net for the three months ended March 31, 2023 included \$183.2 million of a U.S. pension settlement loss. See Note 5 of the Notes to Consolidated Financial Statements for details of other (expense), net.

The loss before income taxes for the three months ended March 31, 2023 was \$154.6 million compared with a loss of \$52.9 million for the three months ended March 31, 2023 was a U.S. pension settlement loss of \$183.2 million.

The provision for income taxes was \$19.9 million for the three months ended March 31, 2023 compared with a provision of \$4.1 million for the three months ended March 31, 2022. The change in the tax provision is a result of the geographic distribution of income as described below.

The company evaluates quarterly the realizability of its deferred tax assets by assessing its valuation allowance and by adjusting such amount, if necessary. The company records a tax provision or benefit for those international subsidiaries that do not have a full valuation allowance against their deferred tax assets. Any profit or loss recorded for the company's U.S. operations will have no provision or benefit associated with it due to the company's valuation allowance, except with respect to refundable tax credits and withholding taxes not creditable against future taxable income. As a result, the company's provision or benefit for taxes may vary significantly period to period depending on the geographic distribution of income.

The realization of the company's net deferred tax assets as of March 31, 2023 is primarily dependent on the ability to generate sustained taxable income in various jurisdictions. Judgment is required to estimate forecasted future taxable income, which may be impacted by future business developments, actual results, strategic operational and tax initiatives, legislative, and other economic factors and developments. Any increase or decrease in the valuation allowance would result in additional or lower income tax expense in that period and could have a significant impact on that period's earnings.

Net loss attributable to Unisys Corporation for the three months ended March 31, 2023 was \$175.4 million, or \$2.58 per diluted share, compared with a loss of \$57.3 million, or \$0.85 per diluted share, for the three months ended March 31, 2022. Included in the loss for the three months ended March 31, 2023 was a U.S. pension settlement loss of \$183.2 million.

Segment results

The company's reportable segments are as follows:

- Digital Workplace Solutions (DWS), which provides modern and traditional workplace solutions;
- Cloud, Applications & Infrastructure Solutions (CA&I), which provides digital platform, applications, and infrastructure solutions; and



• Enterprise Computing Solutions (ECS), which provides solutions that harness secure, continuous high-intensity computing and enable digital services through software-defined operating environments.

The accounting policies of each segment are the same as those followed by the company as a whole. The company evaluates segment performance based on gross profit exclusive of the service cost component of postretirement income or expense, restructuring charges, amortization of purchased intangibles and unusual and nonrecurring items, which are included in other gross profit.

Three months ended March 31, 2023 compared with the three months ended March 31, 2022

A summary of the company's operations by segment is presented below:

	Total S	Segments	DWS	CA&I		ECS	
Three Months Ended March 31, 2023							
Revenue	\$	445.2	\$ 131.0	\$	126.0	\$	188.2
Gross profit percent		35.4 %	 11.9 %		13.0 %		66.7 %
Three Months Ended March 31, 2022							
Revenue	\$	374.5	\$ 124.8	\$	129.1	\$	120.6
Gross profit percent		22.9 %	 12.8 %		5.4 %		52.1 %

DWS revenue was \$131.0 million for the three months ended March 31, 2023, an increase of 5.0% compared with the three months ended March 31, 2022. The increase in revenue was primarily due to recent contract signings. Foreign currency fluctuations had a 3 percentage-point negative impact on DWS revenue in the current period compared with the prior-year period. Gross profit percent was 11.9% in the current period compared with 12.8% in the prior-year period. The decrease in gross profit was primarily due to incremental labor costs in support of recent contract signings.

CA&I revenue was \$126.0 million for the three months ended March 31, 2023, a decline of 2.4% compared with the three months ended March 31, 2022. Foreign currency fluctuations had a 1 percentage-point negative impact on CA&I revenue in the current period compared with the prior-year period. Gross profit percent was 13.0% in the current period compared with 5.4% in the prior-year period. The increase in gross profit was primarily due to additional cost incurred in the prior-year period associated with certain contracts as well as delivery improvements.

ECS revenue was \$188.2 million for the three months ended March 31, 2023, an increase of 56.1% compared with the three months ended March 31, 2022. Foreign currency fluctuations had a 4 percentage-point negative impact on ECS revenue in the current period compared with the prior-year period. Gross profit percent was 66.7% in the current period compared with 52.1% in the prior-year period. The increase in revenue and gross profit was driven by higher software license renewals.

Financial condition

The company's principal sources of liquidity are cash on hand, cash from operations and its revolving credit facility, discussed below. The company and certain international subsidiaries have access to uncommitted lines of credit from various banks. The company believes that it will have adequate sources of liquidity to meet its expected cash requirements for at least the next twelve months.

Cash and cash equivalents at March 31, 2023 were \$391.9 million compared to \$391.8 million at December 31, 2022.

As of March 31, 2023, \$248.9 million of cash and cash equivalents were held by the company's foreign subsidiaries and branches operating outside of the U.S. The company may not be able to readily transfer approximately one-third of these funds out of the country in which they are located as a result of local restrictions, contractual or other legal arrangements or commercial considerations. Additionally, any transfers of these funds to the U.S. in the future may require the company to accrue or pay withholding or other taxes on a portion of the amount transferred.

During the three months ended March 31, 2023, cash provided by operations was \$12.8 million compared to cash usage of \$33.0 million during the three months ended March 31, 2022, primarily driven by higher Technology collections.

Cash used for investing activities during the three months ended March 31, 2023 was \$11.5 million compared to cash usage of \$21.7 million during the three months ended March 31, 2022. Net proceeds of investments were \$9.2 million for the three months ended March 31, 2023 compared with net purchases of \$2.3 million in the prior-year period. Proceeds from investments and purchases of investments represent derivative financial instruments used to reduce the company's currency exposure to market risks from changes in foreign currency exchange rates. In the current period, the investment in marketable software was



\$10.3 million compared with \$11.1 million in the prior-year period, capital additions of properties were \$7.3 million compared with \$5.2 million in the prior-year period and capital additions of outsourcing assets were \$2.7 million compared with \$2.4 million in the prior-year period.

Cash used for financing activities during the three months ended March 31, 2023 was \$7.6 million compared to cash used of \$11.2 million during the three months ended March 31, 2022.

In March 2023, the company purchased a group annuity contract, with plan assets, for approximately \$265 million to transfer projected benefit obligations related to approximately 8,650 retirees of one of the company's U.S. defined benefit pension plans resulting in a pre-tax settlement loss of \$183.2 million for the three months ended March 31, 2023. After considering this most recent group annuity contract purchase, the company has successfully reduced its global defined benefit pension obligations since December 2020 by \$1.7 billion, including \$1.0 billion in the U.S. The company will continue to evaluate opportunities for additional reductions in future periods depending on overall market conditions.

In 2023, the company expects to make cash contributions of approximately \$40 million primarily for its international defined benefit pension plans. In 2022, the company made cash contributions of \$39.3 million to its worldwide defined benefit pension plans. For the three months ended March 31, 2023 and 2022, the company made cash contributions of \$14.5 million and \$15.1 million, respectively.

At the end of each year, the company estimates its future cash contributions to its U.S. qualified defined benefit pension plans based on year-end pension data and assumptions. Any material deterioration in the value of the company's U.S. qualified defined benefit pension plan assets, as well as changes in pension legislation, discount rate changes, asset return changes, or changes in economic or demographic trends, could require the company to make cash contributions to its U.S. qualified defined benefit pension plans in different amounts and on a different schedule than previously contemplated. Based upon our most current estimates as of March 31, 2023, the company does not expect to make mandatory cash contributions to its U.S. qualified defined benefit pension plans until 2025.

At March 31, 2023, total debt was \$506.1 million compared to \$513.1 million at December 31, 2022.

The company has a secured revolving credit facility (the Amended and Restated ABL Credit Facility) that expires on October 29, 2025 that provides for revolving loans and letters of credit up to an aggregate amount of \$145.0 million (with a limit on letters of credit of \$40.0 million), with an accordion feature provision allowing for the aggregate amount available under the credit facility to be increased up to \$175.0 million upon the satisfaction of certain conditions specified in the Amended and Restated ABL Credit Facility. Availability under the credit facility is subject to a borrowing base calculated by reference to the company's receivables. At March 31, 2023, the company had no borrowings and \$6.6 million of letters of credit outstanding, and availability under the facility was \$64.0 million net of letters of credit issued.

The Amended and Restated ABL Credit Facility is subject to a springing maturity, under which the Amended and Restated ABL Credit Facility will immediately mature 91 days prior to any date on which contributions to pension funds in the United States in an amount in excess of \$100.0 million are required to be paid unless the company is able to meet certain conditions, including that the company has the liquidity (as defined in the Amended and Restated ABL Credit Facility) to cash settle the amount of such pension payments, no default or event of default has occurred under the Amended and Restated ABL Credit Facility, the company's liquidity is above \$130.0 million and the company is in compliance with the then applicable fixed charge coverage ratio on a pro forma basis.

The Amended and Restated ABL Credit Facility is guaranteed by Unisys Holding Corporation, Unisys NPL, Inc. and Unisys AP Investment Company I, each of which is a U.S. corporation that is directly or indirectly owned by the company (the subsidiary guarantors). The facility is secured by the assets of the company and the subsidiary guarantors, other than certain excluded assets, under a security agreement entered into by the company and the subsidiary guarantors in favor of JPMorgan Chase Bank, N.A., as agent for the lenders under the credit facility.

The company is required to maintain a minimum fixed charge coverage ratio if the availability under the Amended and Restated ABL Credit Facility falls below the greater of 10% of the lenders' commitments under the facility and \$14.5 million.

The Amended and Restated ABL Credit Facility contains customary representations and warranties, including, but not limited to, that there has been no material adverse change in the company's business, properties, operations or financial condition. The Amended and Restated ABL Credit Facility includes restrictions on the ability of the company and its subsidiaries to, among other things, incur other debt or liens, dispose of assets and make acquisitions, loans and investments, repurchase its equity, and prepay other debt. These restrictions are subject to several important limitations and exceptions. Events of default include non-payment, failure to comply with covenants, materially incorrect representations and warranties, change of control and default under other debt aggregating at least \$50.0 million, subject to relevant cure periods, as applicable.

At March 31, 2023, the company has met all covenants and conditions under its various lending and funding agreements. For at least the next 12 months, the company expects to continue to meet these covenants and conditions.



From time to time, the company may explore a variety of additional debt and equity sources to fund its liquidity and capital needs.

The company may, from time to time, redeem, tender for, or repurchase its securities in the open market or in privately negotiated transactions depending upon availability, market conditions and other factors.

The company does not have any off-balance sheet arrangements that are material or reasonably likely to become material to its financial condition or results of operations.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

There has been no material change in the company's assessment of its sensitivity to market risk since its disclosure in its Annual Report on Form 10-K for the year ended December 31, 2022.

Item 4. Controls and Procedures

Disclosure Controls and Procedures

As of the end of the period covered by this Quarterly Report, management performed, with the participation of the Chief Executive Officer (CEO) and the Chief Financial Officer (CFO), an evaluation of the effectiveness of the company's disclosure controls and procedures as defined in Rules 13a-15(e) and 15d-15(e) of the Securities Exchange Act of 1934 (the Exchange Act). In designing and evaluating the disclosure controls and procedures, management recognized that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives. Based upon that evaluation, the CEO and the CFO concluded that due to material weaknesses in our disclosure controls and procedures and in our internal control over financial reporting, the company's disclosure controls and procedures were not effective as of March 31, 2023 at the reasonable assurance level. A material weakness is a deficiency, or a combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of a company's annual or interim financial statements will not be prevented or detected on a timely basis.

As previously reported in the company's Annual Report on Form 10-K for the year ended December 31, 2022 in connection with the company's assessment of the effectiveness of its internal control over financial reporting at the end of its last fiscal year, management identified the following material weaknesses in the company's disclosure controls and procedures and in the company's internal control over financial reporting as of December 31, 2022 and is in the process of remediating them as of March 31, 2023.

The company did not design and maintain effective formal policies and procedures to ensure appropriate information is communicated from the IT function and the legal and compliance function to the accounting function and those responsible for governance on a timely basis to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. These material weaknesses did not result in a misstatement of the company's financial statements; however, they could have resulted in misstatements of interim or annual consolidated financial statements and disclosures that would result in a material misstatement that would not be prevented or detected.

This section of Item 4, "Controls and Procedures," should be read in conjunction with Item 9A, "Controls and Procedures," included in the company's Annual Report on Form 10-K for the year ended December 31, 2022, for additional information on Management's Report on Internal Control over Financial Reporting.

To address the material weaknesses referenced above, the company performed additional analysis and performed other procedures in order to prepare the unaudited quarterly consolidated financial statements in accordance with generally accepted accounting principles. Accordingly, management believes that the consolidated financial statements included in this Quarterly Report on Form 10-Q present, in all material respects, our financial condition, results of operations and cash flows for the periods presented.

Remediation Plan for Material Weaknesses

Management has implemented measures designed to ensure that the material weaknesses are remediated. The company took the following remediation steps during the fourth quarter of 2022:

- The company enhanced its written policy regarding information escalation for cyber-incidents. In addition, the company completed an assessment of staffing within the company's incident response team.
- The company enhanced its disclosure committee (the Disclosure Committee) and the disclosure working group that supports the Disclosure Committee.



- The company is requiring all direct reports to the CEO to confirm that they have made the Disclosure Committee aware of any matters under their purview that the Disclosure Committee should be considering in advance of applicable SEC filings.
- The company provided training and policies (including any policy revisions) to non-finance executives regarding escalation of significant matters related to SEC reporting requirements.
- Procedures were drafted to address the proper handling of information so that the Security and Risk Committee and Audit and Finance Committee are properly informed.
- Management has revised its Speak Up Policy to make all associates aware that they have direct access to, and may approach, company executives and the Board of Directors, and that they have access to the company's whistleblower hotline.

As of March 31, 2023, management has implemented all remedial actions described above in respect to the material weaknesses relating to policies and procedures within the IT function and the legal and compliance function to the accounting function. Due to the timing of the design and implementation of these remediation efforts implemented during the fourth quarter of 2022, there has been insufficient time for the company to demonstrate consistent execution against all newly implemented actions. As such, management is unable to conclude on the operating effectiveness of implemented remediations at March 31, 2023. We expect to continue to enhance these controls and assess their operating effectiveness throughout 2023.

Changes in Internal Control Over Financial Reporting

Except as described above with respect to our remediation plan, there have been no changes in our internal control over financial reporting during the quarter ended March 31, 2023, that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

Part II - OTHER INFORMATION

Item 1. Legal Proceedings

Information with respect to litigation is set forth in Note 13 of the Notes to Consolidated Financial Statements, and such information is incorporated herein by reference.

Item 1A. Risk Factors

There have been no significant changes to the "Risk Factors" in Part I, Item 1A of the company's Annual Report on Form 10-K for the year ended December 31, 2022.

7	7
4	/

CAUTIONARY STATEMENT PURSUANT TO THE U.S. PRIVATE SECURITIES LITIGATION REFORM ACT OF 1995

Risks and uncertainties that could cause the company's future results to differ materially from those expressed in "forward-looking" statements include:

Implementation of Business Strategy in Information Technology Market

- our ability to grow revenue and expand margin in our Digital Workplace Solutions and Cloud, Applications & Infrastructure Solutions businesses;
- our ability to maintain our installed base and sell new solutions and related services;
- our ability to attract and retain experienced personnel in key positions;
- the potential adverse effects of aggressive competition;
- our ability to effectively anticipate and respond to rapid technological innovation in our industry;
- our ability to retain significant clients and attract new clients;
- our contracts may not be as profitable as expected or provide the expected level of revenues;
- the business and financial risk in implementing acquisitions or dispositions;

Defined Benefit Pension Plans

we have significant underfunded pension obligations;

General Business Risks

- cybersecurity incidents could result in incurring significant costs and harm to our business and reputation;
- our failure to remediate material weaknesses in our disclosure controls and procedures and internal controls over financial reporting or any other material weaknesses in the future could result in material misstatements in our financial statements;
- our ability to access financing markets;
- the risks of doing business internationally when a significant portion of our revenue is derived from international operations;
- the adverse effects of global economic conditions, acts of war, terrorism, natural disasters or the widespread outbreak of infectious diseases;
- a reduction in our credit rating;
- a significant disruption in our IT systems could adversely affect our business and reputation;
- the performance and capabilities of third parties with whom we have commercial relationships;
- if our clients are not satisfied with our services or products, we may face damage to our reputation or legal liability;
- the potential for intellectual property infringement claims to be asserted against our clients or us;
- the possibility that legal proceedings could affect our results of operations or cash flow or may adversely affect our business or reputation;
- a potential impairment of goodwill or intangible assets;
- a failure to meet standards or expectations with respect to our environmental, social and governance practices; and
- our ability to use our net operating loss carryforwards and certain other tax attributes may be limited.

Other factors discussed in this report, although not listed here, also could materially affect our future results.

Item 6. Exhibits

See Exhibit Index

EXHIBIT INDEX

Exhibit Number	Description
3.1	Restated Certificate of Incorporation of Unisys Corporation (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K filed on April 30, 2010)
3.2	Certificate of Amendment of the Restated Certificate of Incorporation of Unisys Corporation (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K filed on April 28, 2011)
3.3	Certificate of Amendment of the Restated Certificate of Incorporation of Unisys Corporation (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K filed on April 28, 2017)
3.4	<u>Bylaws of Unisys Corporation, as amended through December 14, 2022 (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K filed on December 14, 2022)</u>
10.1	Unisys Corporation Savings Plan, as amended and restated effective January 1, 2023
31.1	<u>Certification of Peter A. Altabef required by Rule 13a-14(a) or Rule 15d-14(a)</u>
31.2	<u>Certification of Debra McCann required by Rule 13a-14(a) or Rule 15d-14(a)</u>
32.1	Certification of Peter A. Altabef required by Rule 13a-14(b) or Rule 15d-14(b) and Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350
32.2	Certification of Debra McCann required by Rule 13a-14(b) or Rule 15d-14(b) and Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350
101	The following financial information from Unisys Corporation's Quarterly Report on Form 10-Q for the quarter ended March 31, 2023 formatted in Inline XBRL (Extensible Business Reporting Language) includes: (i) the Consolidated Statements of Income (Loss), (ii) the Consolidated Statements of Comprehensive Income (Loss), (iii) the Consolidated Balance Sheets, (iv) the Consolidated Statements of Cash Flows, (v) the Consolidated Statements of Equity (Deficit), and (vi) Notes to Consolidated Financial Statements
104	Cover page Interactive Data File (formatted as Inline XBRL and contained in Exhibit 101)

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

UNISYS CORPORATION

By: /s/ Debra McCann Debra McCann Executive Vice President and Chief Financial Officer (Principal Financial Officer)

/s/ Erin Mannix Erin Mannix Vice President and Chief Accounting Officer (Principal Accounting Officer)

30

By:

Date: May 2, 2023

UNISYS CORPORATION SAVINGS PLAN

Amended and Restated Effective January 1, 2023

DB1/ 60116903.40

ARTICLE I HISTORY AND SCOPE	1
ARTICLE II DEFINITIONS	3
ARTICLE III ELIGIBILITY FOR PARTICIPATION	15
ARTICLE IV CONTRIBUTIONS	16
ARTICLE V LIMITATIONS ON EMPLOYER CONTRIBUTIONS	22
ARTICLE VI INVESTMENT AND VALUATION OF ACCOUNTS	26
ARTICLE VII VESTING	30
ARTICLE VIII AMOUNT OF BENEFITS	31
ARTICLE IX PAYMENT AND FORM OF BENEFITS	31
ARTICLE X WITHDRAWALS AND LOANS	35
ARTICLE XI SPECIAL PROVISIONS FOR TOP-HEAVY PLANS	40
ARTICLE XII PLAN ADMINISTRATION	41
ARTICLE XIII AMENDMENT AND TERMINATION	48
ARTICLE XIV MISCELLANEOUS	49
APPENDIX A	53

i

UNISYS CORPORATION SAVINGS PLAN

Amended and Restated Effective January 1, 2023

ARTICLE I

HISTORY AND SCOPE

1.01 <u>History</u>. Unisys Corporation (formerly, Burroughs Corporation), adopted the Burroughs Plan, effective July 1, 1984. Unisys Corporation is successor by merger to Sperry Corporation which, prior to such merger, established and maintained the Sperry Plan. Effective April 1, 1988, the Burroughs Plan and Sperry Plan were merged to form the Plan. The Plan is maintained for the benefit of eligible employees of Unisys Corporation and the eligible employees of its subsidiaries that adopt the Plan.

Effective October 1, 1990, the Company's CTIP was merged into the Plan. Effective November 30, 1992, the RIPII was merged into the Plan. Effective March 31, 1996, the RIP was merged into the Plan.

Effective September 16, 2004, the BCC Retirement Plan was merged into the Plan.

This Plan was amended and restated:

- effective January 1, 1998, to bring the Plan into compliance with the Uniformed Services Employment and Reemployment Act of 1994, the Small Business Job Protection Act of 1996, the Taxpayer Relief Act of 1997, the IRS Restructuring and Reform Act of 1998, the Internal Revenue Service Restructuring and Reform Act of 1998, the Community Renewal Tax Relief Act of 2000, and all other applicable law as in effect on the effective date of that amendment and restatement of the Plan.
- effective January 1, 2002, to bring the Plan into compliance with the Economic Growth and Tax Relief Reconciliation Act of 2001, the Job Creation and Worker Assistance Act of 2002, and certain final regulations issued by the Department of Labor and the Department of Treasury.
- effective January 1, 2006, to reflect changes and clarifications related to the administration of the Plan.
- generally effective January 1, 2007, to bring the Plan into compliance with certain final regulations issued under sections 401(k) and 401(m) of the Code, and to reflect certain provisions of the Pension Protection Act of 2006, hurricane relief provisions and certain design changes.
- generally effective January 1, 2008, except as otherwise required by law or provided herein, to add additional
 participating subsidiaries, exclude employees of the Unisys Technical Services division of the Company, and
 to exclude certain paid, nonworking leave from compensation for Plan purposes.
- generally effective January 1, 2010 except as otherwise required by law or provided herein, to reflect certain requirements of the Pension Protection Act of

2006, the Heroes Earnings Assistance and Relief Tax Act of 2008 and the Worker, Retiree and Employer Recovery Act of 2008 and regulations thereunder; and to reflect changes and clarifications related to the administration of the Plan.

- generally effective January 1, 2011 except as otherwise required by law or provided herein, to incorporate amendments through December 31, 2010, and to make certain design changes and clarifications related to the administration of the Plan.
- generally effective January 1, 2012 except as otherwise required by law or provided herein, to reflect certain
 requirements of the Worker, Retiree and Employer Recovery Act of 2008 relative to minimum required
 distributions for 2009, and to make certain design changes and clarifications related to the administration of
 the Plan.

The Plan is now amended and restated effective January 1, 2023 except as otherwise required by law or provided herein, to incorporate prior Plan amendments and make certain design changes related to the Plan participation of Covered Employees of CompuGain LLC and CompuGain Public Services, LLC.

1.02 <u>Effective Dates</u>. The original effective date of the Plan was April 1, 1988. This amendment and restatement of the Plan is generally effective January 1, 2023, except as otherwise required by law or provided herein.

1.03 <u>Rights Affected</u>. Unless provided to the contrary herein or required by law, the provisions of the Plan shall apply to Employees who are credited with an Hour of Service after December 31, 2023.

1.04 <u>Qualification Under the Internal Revenue Code</u>. It is intended that the Plan be a qualified plan within the meaning of section 401(a) of the Code and that the Trust be exempt from federal income taxation under the provisions of section 501(a) of the Code.

1.05 <u>Documents</u>. The Plan consists of the Plan document as set forth herein and any subsequent amendments thereto.

1.06 Plan Merger.

(a) Effective December 30, 2016, the Unisys Technical Services Savings Plan (the 'UTS Plan') and its trust shall be merged with and into the Plan and its Trust. All beneficiary designations and contribution, investment, distribution and withdrawal elections made by a Participant, and in effect on December 30, 2016, under the Unisys Technical Services Savings Plan shall remain in effect under this Plan until changed by the Participant in accordance with the applicable provisions of the Plan. Any outstanding loan taken by a Participant under the UTS Plan shall be repaid under this Plan and shall be taken into account when determining eligibility for a loan under this Plan.

(b) Effective December 31, 2021, the Unify Square, Inc. 401(k) Plan (the "Unify Plan") and its trust shall be merged with and into the Plan and its Trust. All contribution elections in effect on December 31, 2021 under the Unify Plan, and investment elections made by a participant in the Unify Plan prior to the merger with respect to the investment of his Account under the Plan, shall remain in effect under this Plan until changed by the Participant in accordance with the applicable provisions of the Plan; provided, however, that if the Participant's contribution election under the Unify

2 DB1/ 60116903.40 Plan designated a dollar contribution, that dollar contribution level shall be converted to a percentage contribution election. Any beneficiary designation made by a Participant under the Unify Plan shall not be taken into account when determining the Participant's Beneficiary under this Plan. Any outstanding loan taken by a Participant under the Unify Plan shall be repaid under this Plan and shall be taken into account when determining eligibility for a loan under this Plan. Amounts attributable to contributions under the Unify Plan shall be allocated among a Participant's Accounts in accordance with procedures established by the Plan Manager.

1.07 <u>New Employer</u>. Effective February 21, 2022, CompuGain LLC and CompuGain Public Services, LLC (collectively referred to as "CompuGain") shall become Employers under the Plan.

ARTICLE II

DEFINITIONS

The following words and phrases as used herein have the following meanings unless a different meaning is plainly required by the context:

2.01 "<u>Account</u>" means a Participant's After-Tax Account, ESOP Account, GPEP Account, Regular Account, Tax Deferred Account, Roth Contribution Account, Roth In-Plan Conversion Account, Tax Deductible Contribution Account, Qualified Nonelective ESOP Contribution Account, Qualified Nonelective Non-ESOP Contribution Account, Plan Expense Contribution Account, and Rollover Account.

2.02 "<u>Actual Contribution Percentage</u>" means, with respect to a Plan Year, the ratio (expressed as a percentage) of the sum of the amount of (a) Matching Contributions, (b) After-Tax Contributions, (c) Qualified Nonelective ESOP Contributions, and (d) Tax Deferred Contributions recharacterized as After-Tax Contributions, made on behalf of the Participant for the Plan Year to the Participant's Testing Compensation for the Plan Year.

2.03 "<u>Actual Deferral Percentage</u>" means, with respect to a Plan Year, the ratio (expressed as a percentage) of the amount of Tax Deferred Contributions made pursuant to Section 4.01(a) and Qualified Nonelective Non-ESOP Contributions made on behalf of the Participant for the Plan Year to the Participant's Testing Compensation for the Plan Year.

2.04 "<u>Administrative Committee</u>" means the committee appointed in accordance with Section 12.02, which is responsible for reviewing and deciding appeals under the Plan.

2.05 "<u>Affiliate</u>" means any entity included with the Employer in (a) a controlled group of employers or trades or businesses within the meaning of section 414(b) or 414(c) of the Code; (b) an affiliated service group within the meaning of section 414(m) of the Code; or (c) a group required to be aggregated pursuant to the regulations under section 414(o) of the Code; provided that any such employer shall be included within the term "Affiliate" only while a member of a group including the Employer. For purposes of Section 5.04, whether a member of a controlled group is an Affiliate shall be determined under section 1563(a) of the Code (as incorporated through application of sections 414(b) and (c) of the Code) by substituting "50%" for "80%" everywhere it appears in section 1563(a) of the Code.

3 DB1/ 60116903.40 2.06 "<u>After-Tax Account</u>" means a Participant's account to which are credited After-Tax Contributions, if any, and earnings and losses thereon.

2.07 "<u>After-Tax Contribution</u>" means a contribution made by an Employee in accordance with a Participant's salary reduction agreement pursuant to Section 4.02(b).

2.08 "Aggregation Group" means the group of qualified plans sponsored by the Employer or by an Affiliate formed by including in such group (a) all such plans in which a Key Employee participates in the Plan Year containing the Determination Date, or any of the four preceding Plan Years, including any frozen or terminated plan that was maintained within the five-year period ending on the Determination Date, (b) all such plans which enable any plan described in clause (a) to meet the requirements of either section 401(a)(4) of the Code or section 410 of the Code, and (c) such other qualified plans sponsored by the Employer or an Affiliate as the Employer elects to include in such group, as long as the group, including those plans electively included, continues to meet the requirements of sections 401(a)(4) and 410 of the Code.

2.09 "BCC" means Baesch Computer Consulting.

2.10 "Beneficiary" means (a) the Participant's Spouse, or (b) the person, persons or trust designated by the Participant, in writing on forms provided by the Plan or in accordance with procedures established by the Plan, with the consent of his Spouse, if any, as direct or contingent beneficiary. In order to be valid, the Spouse's consent to a Beneficiary other than or in addition to the Participant's Spouse, must be in writing on forms provided by the Plan or in accordance with procedures established by the Plan, must consent to the specific Beneficiary designated, must acknowledge the effect of such consent, and must be witnessed by a Plan representative or notary public. If the Participant has no Spouse and no effective beneficiary designation, his Beneficiary shall be the first of the following classes in which there is any person surviving the Participant: (a) the Participant's children, (b) the Participant's parents, and (c) the Participant's brothers and sisters. Unless otherwise provided in the applicable Beneficiary form, if the Participant has no spouse, if none of the foregoing classes include a person surviving the Participant, the Participant's Beneficiary shall be his estate. For the avoidance of doubt, any beneficiary designation completed by a Participant under the Unify Plan shall not apply under the Plan.

2.11 "<u>Benefit Commencement Date</u>" means the first day on which all events have occurred that entitle a Participant to the benefit.

2.12A "<u>Benefit Strategy Committee</u>" means the Plan fiduciary described in Article IX with respect to the administration of, and the control and management of Investment Funds under the Plan, except to the extent that fiduciary authority over (a) the control and management of Investment Funds under the Plan has been delegated to the Investment Committee, (b) Plan administration has been delegated to the Plan Manager and (c) benefit claims has been delegated to the Administrative Committee.

2.12 "Board" means the Board of Directors of the Company.

2.13 "Burroughs Plan" means the Burroughs Employees Savings Thrift Plan, as in effect on March 30, 1988.

- 2.14 "Code" means the Internal Revenue Code of 1986, as amended.
- 2.15 "Company" means Unisys Corporation.

"Compensation" means a Participant's wages or salary paid by an Employer to an Employee, including 2.16 amounts deducted in accordance with section 125, 132(f)(4) or 401(k) of the Code, overtime pay, shift differentials, overseas hardship and war risk premiums, payments for accrued but unused vacation, commissions paid under the terms of a written ongoing sales commission plan, paid bonuses paid under the terms of a written ongoing bonus plan approved as such by the Plan Manager, military differential wage payments made by the Employer to a Participant in accordance with section 3401(h) and section 414(u)(12) of the Code and Workers' Compensation differential wage payments made by the Employer to a Participant, but excluding any amounts received by an Employee while he is not a Participant, additional fringe benefit payments related to the Service Contract Act, any other deferred compensation, "garden leave payments," California additional short-term disability, family medical leave and meal premium payments and penalties, bonuses paid under the ONELead Program (and similar bonus programs) and amounts in excess of the dollar limitation in effect under section 401(a)(17) of the Code with respect to any Plan Year, and any amounts that are excluded from the definition of compensation set forth in section 415(c)(3) of the Code. Notwithstanding the foregoing, any amounts deducted on a pre-tax basis for group health coverage because the Participant is unable to certify that he or she has other health coverage, so long as the Employer does not otherwise request or collect information regarding the Participant's other health coverage as part of the enrollment process for the Employer's health plan, shall be included as Compensation.

For purposes of this Section 2.16, "garden leave payments" are certain amounts negotiated under a Participant's termination agreement that are paid during periods when no services are performed by such Participant.

2.17 "Covered Employee" means any Employee other than:

(a) any Employee who is a member of a collective bargaining unit, unless such collective bargaining agreement provides for the Employee's participation in the Plan;

(b) any Employee who is a nonresident alien of the United States (including the District of Columbia or the Virgin Islands) and who does not receive any United States (including the District of Columbia or the Virgin Islands) source income from the Employer;

(c) an Employee who is (1) employed by an overseas subsidiary of an Employer and (2) on temporary assignment to the Employer;

(d) any individual who is not an employee of the Employer but who provides services as described in section 414(n)(2) of the Code;

(e) any individual who is classified as an independent contractor by the Employer or any persons who are not treated by the Employer as employees for purposes of withholding federal employment taxes, regardless of (1) how such individual is classified by the Internal Revenue Service, other governmental agency, government or court, or (2) a contrary governmental or judicial determination relating to such employment status or tax withholding;

(f) between September 26, 2006 and January 1, 2017, an Employee who is employed by Unisys Technical Services L.L.C.;

(g) between January 1, 2008 and January 1, 2017, an Employee who is employed by the Unisys Technical Services division of the Company; and

(h) effective December 31, 2012, an Employee who is covered by the Unisys Savings Plan for Puerto Rico Employees.

2.18 "CTIP" means the Convergent Tax Investment Plan, as in effect on September 30, 1990.

2.19 "Determination Date" means the last day of the preceding Plan Year.

2.20 "<u>Distributee</u>" means a Participant, the surviving Spouse of a deceased Participant, or a Participant's Spouse or former Spouse who is an alternate payee under a Qualified Domestic Relations Order.

2.21 "<u>Employee</u>" means (a) an individual who is employed by the Employer, (b) when required by context for purposes of crediting Hours of Service under Section 2.29, a former Employee, and (c) a leased employee as described under section 414(n)(2) of the Code.

2.22 "<u>Employer</u>" means the Company and any Affiliate that, with the consent of (a) the Company, or (b) the Chief Executive Officer of the Company (or an entity or individual to whom such responsibility has been delegated), adopts the Plan and joins in the Trust Agreement.

2.23 "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

2.24 "<u>ESOP Account</u>" means a Participant's account to which are credited Matching Contributions made to the Plan after March 31, 1989 and which were contributed, or absent the Participant's election, invested in shares of Unisys Stock, and earnings and losses thereon.

2.25 "ESOP Portion of the Plan" means the portion of the Plan that is both a stock bonus plan and an employee stock ownership plan intended to qualify under sections 401(a) and 4975(e)(7) of the Code, the assets of which are held in the ESOP Account and Qualified Nonelective ESOP Accounts of Participants and invested primarily in shares of Unisys Stock that meet the requirements of section 409(I) of the Code.

2.26 "<u>Fund</u>" means the assets and all earnings, appreciation and additions thereto, less losses, depreciation and any proper payments made by the Trustee, held under the Trust by the Trustee for the exclusive benefit of Participants and their Beneficiaries.

2.27 "<u>GPEP Account</u>" means a Participant's account to which are credited GPEP contributions made with respect to Plan Years beginning before January 1, 1998, if any, and earnings and losses thereon.

2.28 "Highly Compensated Employee" means an Employee who either:

(a) was a 5% owner (as defined in section 416(i)(1) of the Code) at any time during the Plan Year for which Highly Compensated Employees are being identified or the preceding Plan Year; or

(b) with respect to the Plan Year preceding the calendar year for which Highly Compensated Employees are being identified both (1) had Testing Compensation in excess of the dollar amount under section 414(q)(1)(B)(i) of the Code, as in effect for such Plan Year, and (2) was in the top 20% of all Employees when ranked on the basis of Testing Compensation.

2.29 "<u>Hour of Service</u>" means each hour for which an Employee is directly or indirectly paid or entitled to payment by the Company or an Affiliate for the performance of duties.

2.30 <u>"Investment Committee</u>" means the Pension Investment Review Committee appointed pursuant to Section 12.02 which is responsible for the control and management of the Investment Funds.

2.31 "<u>Investment Fund</u>" means a fund selected by the Investment Committee in which the Fund or any portion thereof may be invested.

2.32 "<u>Investment Manager</u>" means the individual or entity, if any, selected by the Trustee responsible for the investment of all or a portion of the Fund.

2.33 "Key Employee" means a person employed or formerly employed by the Employer or an Affiliate who, during the Plan Year, was any of the following:

(a) an officer of the Employer having annual Testing Compensation of more than 130,000, or such other amount as may be in effect under section 415(1)(A)(i) of the Code;

(b) a 5% owner of the Employer.

(c) a person who is both an employee whose annual Testing Compensation exceeds \$150,000 and who is a 5% owner of the Employer.

The Beneficiary of any deceased Participant who was a Key Employee shall be considered a Key Employee for the same period as the deceased Participant would have been so considered.

2.34 "Key Employee Ratio" means the ratio (expressed as a percentage) for any Plan Year, calculated as of the Determination Date with respect to such Plan Year, determined by dividing the amount described in subsection (a) hereof by the amount described in subsection (b) hereof, after deduction from both such amounts of the amount described in subsection (c) hereof.

(a) The amount described in this subsection (a) is the sum of (1) the aggregate of the present value of all accrued benefits of Key Employees under all qualified defined benefit plans included in the Aggregation Group, (2) the aggregate of the balances in all of the accounts standing to the credit of Key Employees under all qualified defined contribution plans included in the Aggregation Group, and (3) the aggregate amount distributed from all plans in such Aggregation Group to or on behalf of any Key Employee during the one-year period ending on the Determination Date. In the case of a distribution made for a reason other than severance from employment, death, or disability, clause (3) herein shall be applied by substituting "five-year period" for "one-year period."

(b) The amount described in this subsection (b) is the sum of (1) the aggregate of the present value of all accrued benefits of all Participants under all qualified defined benefit plans included in the Aggregation Group, (2) the aggregate of the balances in all of the accounts standing to the credit of all Participants under all qualified defined contribution plans included in the Aggregation Group, and (3) the aggregate amount distributed from all plans in such Aggregation Group to or on behalf of any Participant during the one-year period ending on the Determination Date. In the case of a distribution made for a reason other than severance from employment, death, or disability, clause (3) herein shall be applied by substituting 'five-year period' for 'one-year period.'

(c) The amount described in this subsection (c) is the sum of (1) all rollover contributions (or similar transfers) to plans included in the Aggregation Group initiated by an Employee from a plan sponsored by an employer which is not the Employer or an Affiliate, (2) any amount that would have been included under subsection (a) or (b) hereof with respect to any person who has not rendered service to any Employer at any time during the one-year period ending on the Determination Date, and (3) any amount that is included in subsection (b) hereof for, on behalf of, or on account of, a person who is a Non Key Employee as to the Plan Year of reference but who was a Key Employee as to any earlier Plan Year.

The present value of accrued benefits under any defined benefit plan shall be determined under the method used for accrual purposes for all plans maintained by the Employer and all Affiliates if a single method is used by all such plans, or otherwise, the slowest accrual method permitted under section 411(b)(1)(C) of the Code.

2.35 "<u>Matching Contribution</u>" means a contribution made by an Employer in accordance with Section 4.03.

2.36 "<u>Matching Contribution Account</u>" means a Participant's account to which are credited Matching Contributions made to the Plan after March 31, 1989 and which were not contributed, or absent the Participant's election, invested in shares of Unisys Stock, and earnings and losses thereon.

2.37 "<u>Non-Highly Compensated Employee</u>" means an Employee other than a Highly Compensated Employee.

2.38 "<u>Non-Key Employee</u>" means any Employee or former Employee who is not a Key Employee as to that Plan Year, or a Beneficiary of a deceased Participant who was a Non-Key Employee.

2.39 "Normal Retirement Age" means age 65.

2.40 "<u>Notice Period</u>" means the period beginning 90 days before and ending 30 days before the Benefit Commencement Date.

2.41 "<u>Participant</u>" means a Covered Employee who has met the eligibility requirements of Section 3.01. An individual who is a Participant but who ceases to be a Covered Employee shall nonetheless remain a Participant for purposes of benefit payments only, until all amounts due him under the Plan have been paid.

2.42 "<u>Period of Severance</u>" means a period beginning on the date of an Employee's Severance from Employment and ending on the date on which the Employee again performs an Hour of Service.

Notwithstanding the foregoing, solely for the purpose of determining whether a Period of Severance has occurred, in the case of an absence from employment by reason of the pregnancy of the Employee, the birth of a child of the Employee, the placement of a child with the Employee in connection with the adoption of the child by the Employee or the caring for the child for a period beginning immediately following that birth or placement, the period between the first and second anniversary of the first day of such absence from employment shall neither be construed as a Period of Severance nor a period of Service. In order for an absence to be considered to be for the reasons described in the foregoing sentence, an Employee shall provide the Plan Manager with information regarding the reasons for the absence. Nothing in this Section 2.42 shall be construed as expanding or amending any maternity or paternity leave policy of an Employer or Affiliate.

2.43 "<u>Plan</u>" means the profit sharing plan, known as the "Unisys Savings Plan" set forth in this document, which includes a stock bonus plan and employee stock ownership plan intended to qualify under sections 401(a) and 4975(e)(7) of the Code, and the related trust agreement pursuant to which the Trust is maintained.

2.44 "<u>Plan Expense Contribution</u>" means a contribution made by an Employer in accordance with Section 4.11.

2.45 "<u>Plan Expense Contribution Account</u>" means a Participant's account to which are credited Plan Expense Contributions and earnings and losses thereon and against which shall be charged Plan expenses as determined by the Plan Manager.

2.46 "<u>Plan Manager</u>" means the individual or individuals responsible for certain matters relating to the administration of the Plan, as described under Article XII.

2.47 "Plan Year" means the calendar year.

2.48 "Prior Plan" means the Burroughs Plan, Sperry Plan, CTIP, RIP, RIPII or BCC Retirement Plan.

2.49 "<u>Qualified Default Investment Alternative</u>" means the Fidelity Freedom Fund closest to the year of the Participant's 65th birthday.

2.50 "<u>Qualified Domestic Relations Order</u>" means a judgment, decree or order that relates to a Participant's benefit under the Plan and meets the requirements of section 414(p) of the Code.

2.51 "<u>Qualified Nonelective ESOP Account</u>" means a Participant's account to which are credited Qualified Nonelective ESOP Contributions, if any, and earnings and losses thereon.

2.52 "<u>Qualified Nonelective ESOP Contribution</u>" means a contribution made by the Employer pursuant to Section 4.05 for purposes of satisfying the requirements of Section 5.03.

2.53 "<u>Qualified Nonelective Non-ESOP Account</u>" means a Participant's Account to which are credited Qualified Nonelective Non-ESOP Contributions, if any, and earnings and losses thereon.

2.54 "<u>Qualified Nonelective Non-ESOP Contribution</u>" means a contribution made by the Employer pursuant to Section 4.05 for purposes of satisfying the requirements of Section 5.02.

2.55 "<u>Regular Account</u>" means a Participant's Account to which are credited (a) Matching Contributions made before April 1, 1989, (b) matching contributions made to a Prior Plan (other than CTIP) before April 1, 1989, (c) matching contributions made to the CTIP before October 1, 1990, (d) employee contributions made to the Sperry Plan, and (e) earnings and losses.

2.56 "<u>RIP</u>" means the Unisys Retirement Investment Plan, as in effect on March 31, 1996.

2.57 "<u>RIPII</u>" means the Retirement Investment Plan II, as in effect on November 30, 1992.

2.58 "<u>Rollover Account</u>" means a Participant's account to which are credited the (a) Participant's Rollover Contributions, if any, (b) amounts, if any, transferred to a Participant's Account from a Prior Plan which were derived from such Participant's rollover contributions to such Prior Plan, and (c) earnings and losses thereon; provided, however, that any portion of a Rollover Contribution consisting of amounts attributable to Roth contributions shall not be allocated to a Participant's Rollover Account, but shall be allocated to a sub-account of the Participant's Roth Contribution Account.

2.59 "<u>Rollover Contribution</u>" means a contribution made by a Participant pursuant to Section 4.06.

2.60A "<u>Roth Contribution Account</u>" means the sub-account of a Participant's Tax Deferred Account to which are credited the (a) Participant's Roth Contributions, if any, and (b) earnings and losses thereon.

2.60B "<u>Roth Contributions</u>" means the portion of a Participant's Tax Deferred Contributions under Sections 4.01(a) and (b) for any period beginning on or after January 1, 2020, if any, that the Participant designates as Roth contributions under section 402A of the Code. Except as specifically provided by federal law, or in the Plan or an amendment thereto, Roth Contributions shall be treated as Salary Deferrals under the Plan.

2.60C "<u>Roth In-Plan Conversion Account</u>" means the sub-account of a Participant's Tax Deferred Account consisting of vested amounts from a Participant's Account that are converted to Roth Contributions in accordance with the provisions of Section 4.12 of the Plan, and section 402A(c)(4) of the Code and the regulations and rulings promulgated thereunder.

2.60 "<u>Service</u>" means the periods determined in accordance with the following provisions of this Section 2.60.

For purposes of determining a Participant's vested interest in his Matching Contribution Account under Section 7.01(c), the Participant's Service shall be determined from the first date the Employee performs an Hour of Service (or, with respect to an Employee who has incurred a Period of Severance, the first date following the Period of Severance on which that Employee performs an Hour of Service) and ending on the date the Employee incurs a Severance from Employment. Years of Service shall be calculated on the basis that 365 days of employment equal one year

and nonconsecutive periods of employment that are not disregarded under subsection (3) below shall be aggregated. The following additional rules shall be in calculating Service under this Section:

(a) If an Employee retires, quits or is discharged or otherwise experiences a Severance from Employment, the period commencing on the Employee's Severance from Employment date and ending on the first date on which he again performs an Hour of Service shall be taken into account in determining the Employee's Service, if such date is within 12 consecutive months of the date on which he last performed an Hour of Service.

(b) If the Employee is absent from work for a reason other than one specified in Section 2.60(a)(1) and within 12 months of the first day of such absence, the Employee retires, quits or is discharged or otherwise experiences a Severance from Employment, the period commencing on the first day of such absence and ending on the first date on which he again performs an Hour of Service shall be taken into account, if such date is within 12 consecutive months of the date on which his absence began.

(c) The Service completed by a Participant or Employee before he incurred a Period of Severance shall be canceled unless the Participant or Employee (1) had a vested interest in his Account (other than his After-Tax Account or his Rollover Account) or (2) had no vested interest in any of his Accounts (other than his After-Tax Account and his Rollover Account) at the time he incurred a Period of Severance and he again completes an Hour of Service before he incurs five consecutive twelve-month Periods of Severance.

(d) Service shall exclude service prior to the date on which a business is acquired, merged, consolidated, or otherwise absorbed by the Company or an Affiliate, or prior to the date the assets of a business are acquired by the Company or an Affiliate, unless otherwise provided herein or authorized by the Company.

2.61 "Severance from Employment" means the earlier of (a) the date an Employee dies or retires, quits or is discharged from the Employer and all Affiliates, or (b) the first anniversary of the date that the Employee is otherwise first absent from work from the Employer and all Affiliates (with or without pay) for any reason; provided, however, that if the Employee's absence is attributable to qualified military service, the Employee shall not be considered to have had a Severance from Employment provided the absent Employee returns to active employment with the Employer or Affiliate.

2.62 "Sperry Plan" means the Sperry Retirement Program - Part B, as in effect on March 30, 1988.

2.63 "Spouse" means:

(a) for periods beginning on and after June 26, 2013, the person to whom the Participant was legally married on the date on which the Participant's marital status must be determined under the applicable Plan provision.

(b) for periods prior to June 26, 2013, the spouse or surviving spouse of the Participant who is a person of the opposite gender who is the lawful husband or lawful wife of a Participant under the laws of the state or country of the Participant 's domicile.

Notwithstanding the foregoing, a former spouse shall be treated as the Spouse or surviving Spouse to the extent provided under a Qualified Domestic Relations Order.

2.64 "<u>Tax Deductible Contribution Account</u>" means a Participant's account to which are credited tax deductible contributions, if any, made to the Plan before April 1, 1989, and earnings and losses thereon.

2.65 "<u>Tax Deferred Account</u>" means a Participant's account to which are credited (a) Tax-Deferred Contributions, if any, (b) tax deferred contributions made under a Prior Plan and transferred to the Plan, (c) basic member contributions, if any, made under the Sperry Plan and transferred to the Plan, and (d) earnings and losses thereon.

2.66 <u>"Tax Deferred Contribution</u>" means a contribution made by an Employer in accordance with a Participant's salary reduction agreement pursuant to Section 4.01(a) and (b). Except as specifically provided by federal law, or in the Plan or an amendment thereto, Tax Deferred Contributions shall include Roth Contributions.

2.67 <u>"Termination of Employment</u>" means an Employee's cessation of employment with the Company and all Affiliates and Associated Companies as a result of quitting, retirement, discharge, release or placement on extended lay-off with no expectation of recall, or failure to return to active employment upon expiration of an approved leave of absence.

"Testing Compensation" means the total of a Participant's wages, salary and other amounts paid by an 2.68 Employer and reported in Internal Revenue Service Form W-2, and any amounts deferred under section 402(g)(3) or 125 of the Code and, effective January 1, 2001, section 132(f)(4) of the Code; provided, however, for purposes of Sections 5.02, 5.03 and 5.04, the Plan Manager may elect to exclude amounts deducted in accordance with sections 125, 132(f)(4), and 402(e)(3) of the Code as Testing Compensation. Notwithstanding the foregoing, any amounts deducted on a pre-tax basis for group health coverage because the Participant is unable to certify that he or she has other health coverage, so long as the Employer does not otherwise request or collect information regarding the Participant's other health coverage as part of the enrollment process for the Employer's health plan, shall be included as Testing Compensation. Compensation for purposes of this Section shall include regular pay as described in Treasury Regulation section 1.415(c)-(2)(e)(3)(ii) if paid by the end of the Limitation Year that includes the Employee's termination of employment, or if later, $2\frac{1}{2}$ months after the Employee's termination of employment ("the Post-Termination Period"). Any payments not described in the foregoing sentence shall not be considered Compensation if paid after termination of employment, even if they are paid within the Post Termination Period. Only the first \$230,000, as adjusted in accordance with section 401(a)(17)(B) of the Code and the regulations thereunder, of the amount otherwise described in this Section shall be counted on or after January 1, 2008. Effective January 1, 2009. Testing Compensation shall include the amount of any military differential wage payments made by the Employer to a Participant in accordance with section 3401(h) and section 414(u)(12) of the Code.

2.69 "<u>Trust</u>" means the legal entity created by the trust agreement between the Employer and the Trustee, fixing the rights and liabilities with respect to controlling and managing the Fund for the purposes of the Plan.

2.70 "<u>Trust Agreement</u>" means the trust agreement between the Company and the Trustee, fixing the rights and liabilities with respect to controlling and managing the Fund for the purposes of the Plan.

2.71 "<u>Trustee</u>" means the party or parties appointed by the Investment Committee as trustee of the Trust and named as trustee pursuant to the Trust Agreement or any successors thereto.

2.73A "Unify Plan" means the Unify Square, Inc. 401(k) Plan as in effect on the applicable date.

2.72 "<u>Unisys Stock</u>" means Unisys Corporation common stock, par value \$0.01 per share, which is readily tradable on an established securities market.

2.73 "<u>Valuation Date</u>" means each day of each calendar year.

ARTICLE III

ELIGIBILITY FOR PARTICIPATION

3.01 <u>Eligibility Requirement</u>. An Employee shall be eligible to become a Participant if he is a Covered Employee.

3.02 <u>Participation Commencement Date</u>. Each Covered Employee who was a Participant as of December 31, 2022, shall continue to be a Participant on January 1, 2023, if he is then a Covered Employee. Each other Covered Employee shall be a Participant on his first day of employment as a Covered Employee.

3.03 <u>Time of Participation-Excluded Employees</u>. An Employee who is ineligible to be a Participant because he is not a Covered Employee, shall become a Participant as of the first day on which he becomes a Covered Employee. A Participant shall cease to be an active Participant on any date on which he ceases to be a Covered Employee; however, a Participant who ceases to be a Covered Employee will remain a Participant for distribution purposes under the Plan until such time as he no longer has a vested interest under the Plan.

ARTICLE IV

CONTRIBUTIONS

4.01 Tax Deferred Contributions.

(a) Subject to the limitations contained in Article V, each Employer shall make a Tax Deferred Contribution for the Plan Year to the Tax Deferred Account of each of its Covered Employees who, with respect to such Plan Year is a Participant and has filed a salary reduction notice with the Employer that provides for a reduction in Compensation otherwise payable to the Participant by a designated whole percentage that does not exceed the limit described in paragraph (2), and a contribution of that amount by the Employer to the Participant's Tax Deferred Account.

(2) The amount of the Tax Deferred Contribution made for a Participant with respect to any Plan Year pursuant to this subsection (a) shall be the amount specified in the salary reduction notice. The percentage specified shall be a whole

percentage of the Participant's Compensation not to exceed (A) 80% (30% for periods prior to January 1, 2016) with respect to a Participant who is a Non-Highly Compensated Employee or (B) 9% (18% for periods prior to January 1, 2017) with respect to a Participant who is a Highly Compensated Employee, The Plan Manager may, in its discretion, increase or decrease the maximum permissible amount of Tax Deferred Contributions at any time and from time to time as it deems appropriate. Any salary reduction notice shall relate only to Compensation as yet unearned when the notice is filed and may not be amended during the period to which it pertains, except that it may be terminated as to amounts unearned at the date of a Participant's Termination of Employment.

(3) A Participant may designate as Roth Contributions any portion of his Tax Deferred Contributions under this subsection (a) for any period beginning on or after January 1, 2020; provided, however, that a Participant employed by CompuGain may not designate as Roth Contributions any portion of his Tax Deferred Contributions between February 21, 2022 and December 31, 2022.

(b) Each Employer shall make an additional Salary Deferral Contribution for the Plan Year to the Tax Deferred Account of each of its Covered Employees who, with respect to such Plan Year is a Participant, is age 50 or older as of the last day of the Plan Year, and has elected, in accordance with procedures established by the Plan Manager and subject to any limitations imposed by the Plan Manager, to make an additional Salary Deferral Contribution in an amount not to exceed \$1,000 for the Plan Year (or such other amount as may be applicable under section 414(v) of the Code), reduced by, to the extent required by the Code and applicable Treasury regulations, any other elective deferrals contributed on the Participant's behalf pursuant to section 414(v) of the Code for the Plan Year; provided, however, that elective deferrals shall be treated for all Plan purposes as contributed under subsection (a) above in lieu of this subsection, unless the Participant is unable to make additional Salary Deferral Contributions under subsection (a) above for the Plan Year due to limitations imposed by the Plan or applicable federal law. A Participant may designate as Roth Contributions any portion of his Tax Deferred Contributions under this subsection (b) for any period beginning on or after January 1, 2020. The amount of the additional Salary Deferral Contributions and/or Roth Contributions elected by the Participant shall be a whole percentage of the Participant's Compensation up to 100% of the Participant's Compensation.

(c) Salary reduction notices pursuant to this Section 4.01 must be made within the time prescribed by the Plan Manager and shall become effective in accordance with the rules and procedures established by the Plan Manager.

(d) Subject to, and in accordance with, the rules and procedures established by the Plan Manager, a Participant may elect to change, discontinue, or resume the percentage of Compensation under his salary reduction notice. All such elections shall become effective in accordance with the rules and procedures established by the Plan Manager.

4.02 After-Tax Contributions.

(a) A Participant may make After-Tax Contributions to the Plan by filing a salary reduction notice authorizing the Employer to reduce the after-tax Compensation otherwise payable to the Participant by a designated whole percentage (up to the limit specified in subsection (b)), and deposit such amounts into the Participant's After-Tax Contribution Account; provided, however, that a Participant employed by CompuGain

may not make After-Tax Contributions to the Plan between February 21, 2022 and December 31, 2022.

(b) The amount of the After-Tax Contribution made by a Participant with respect to any Plan Year shall be the amount specified in the salary reduction notice. The percentage specified shall be a whole percentage not to exceed 6% of the Participant's Compensation.

Any salary reduction notice shall relate only to Compensation as yet unearned when the notice is filed and may not be amended during the period to which it pertains, except that it may be terminated as to amounts unearned at the date of a Participant's Termination of Employment.

(c) Salary reduction notices pursuant to this Section 4.02 must be made within the time prescribed by the Plan Manager and shall become effective in accordance with the rules and procedures established by the Plan Manager.

(d) Subject to, and in accordance with, the rules and procedures established by the Plan Manager, a Participant may elect to change, discontinue, or resume the percentage of Compensation under his salary reduction notice. All such elections shall become effective in accordance with the rules and procedures established by the Plan Manager.

4.03 <u>Matching Contributions</u>. Subject to the limitations in Article V, each Employer may make a Matching Contribution for each Plan Year to the Account of each of its Covered Employees who, with respect to such Plan Year, is a Participant and has filed a salary reduction notice in accordance with Section 4.01. If Matching Contributions are made under the Plan, such Matching Contributions shall be in an amount determined in accordance with subsections (a) and (b) below; provided, however, that no Matching Contribution shall be made for a Participant with respect to any period between February 21, 2022 and December 31, 2022 during which the Participant was employed by CompuGain.

(a) Subject to the minimum set forth in subsection (b),

(1) With respect to a Participant whose employment is not subject to a collective bargaining agreement or whose collective bargaining agreement provides that such Participant shall be treated in the same manner as a non-union Employee, the amount of the Matching Contribution made in accordance with this Section 4.03 with respect to each pay period in the Plan Year commencing January 1, 2011 shall be an amount equal to 50% of the first 6% of Compensation contributed as a Tax Deferred Contribution made pursuant to Section 4.01(a); provided, that the maximum Matching Contribution payable to a Participant shall not equal more than 3% of such Participant's Compensation for the period. With respect to each pay period in the Plan Year commencing January 1, 2007 and prior to January 1, 2009 the Matching Contribution made in accordance with this Section 4.03 shall be an amount equal to 100% of the first 6% of Compensation contributed as a Tax Deferred Contribution made pursuant to Section 4.03 shall be an amount equal to 100% of the first 6% of Compensation contributed as a Tax Deferred Contribution made pursuant to Section 4.01(a); provided, that the maximum Matching Contribution payable to a Participant shall not equal more than 6% of such Participant's Compensation for the period. No Matching Contribution shall be made on or after January 1, 2009 and prior to January 1, 2011.

(2) Effective January 1, 2011, in addition to the Matching Contribution made pursuant to subsection (a)(1) above for a Participant whose

employment is not subject to a collective bargaining agreement or whose collective bargaining agreement provides that such Participant shall be treated in the same manner as a non-union Employee, each Employer shall contribute to the Plan at or following the end of the Plan Year an additional Matching Contribution (a "True-Up Matching Contribution") in an amount equal to (A) 50% of the first 6% of Compensation contributed, by the Participant, as a Tax Deferred Contribution pursuant to Section 4.01(a) for the Plan Year, minus (B) any Matching Contributions previously contributed by the Employer on behalf of such Participant for the Plan Year pursuant to subsection (a)(1) above; provided, however, that, no such True-Up Matching Contribution shall be made on behalf of any Participant who incurred a Termination of Employment during the Plan Year.

(3) With respect to a Participant not described in Section 4.03(a)(1), for Plan Years commencing prior to January 1, 2009, the amount of the Matching Contribution made in accordance with this Section 4.03 with respect to each pay period in the Plan Year shall be an amount equal to 50% of the first 4% of Compensation contributed as a Tax Deferred Contribution made pursuant to Section 4.01(a); provided, that the maximum Matching Contribution payable to a Participant shall not equal more than 2% of such Participant's Compensation for the period. No Matching Contribution shall be made on or after January 1, 2009.

(b) Notwithstanding anything in subsection (a) to the contrary:

(1) each Participant who was employed by an Employer at any time during the period beginning July 1, 1998 and ending December 31, 1998 who had Tax Deferred Contributions made on his behalf for the Plan Year ending December 31, 1998 shall receive a minimum Matching Contribution for such Plan Year in an amount equal to the lesser of:

(A) 1% of the Participant's Compensation not in excess of \$80,000 for the period July 1, 1998 through December 31, 1998; or

(B) 25% of the total of the Tax Deferred Contributions made on behalf of the Participant for the Plan Year (regardless of when the Tax Deferred Contributions were made during such Plan Year).

(2) for periods on or after January 1, 1999 but prior to January 1, 2009, each Participant who was employed by an Employer on December 31 of a Plan Year beginning on or after January 1, 1999 and who had Tax Deferred Contributions made on his behalf shall receive a minimum Matching Contribution, in accordance with procedures adopted by the Plan Manager, in an amount, when added to the Matching Contributions made on behalf of such Participant (before application of this paragraph), equal to (a) in the case of a Participant whose employment is not subject to a collective bargaining agreement or whose collective bargaining agreement provides that such Participant shall be treated in the same manner as a non-union Employee, 6% of the Participant's Compensation not in excess of the limit described in section 401(a)(17) of the Code as in effect with respect to such Plan Year, or (b) in the case of a Participant not described in the preceding subsection (a), the lesser of:

(A) 2% of the Participant's Compensation not in excess of the limit described in section 401(a)(17) of the Code as in effect with respect to such Plan Year; or

the Plan Year.

(B) 50% of the total of the Tax Deferred Contributions made on behalf of the Participant for

(c) The portion of a Participant's Tax Deferred Contributions designated as Roth Contributions shall be taken into account in determining the amount of a Participant's Matching Contributions.

4.04 <u>GPEP Contributions</u>. No contributions may be made to an individual's GPEP Account with respect to any Plan Year beginning on or after January 1, 1998. Amounts, if any, allocated to a Participant's GPEP Account prior to January 1, 1998 shall continue to be held in the GPEP Account until distributed in accordance with the terms of the Plan.

4.05 <u>Qualified Nonelective Contributions</u>. Subject to the limitations described in Article V, each Employer shall make a Qualified Nonelective Non-ESOP Contribution, a Qualified Nonelective ESOP Contribution, or both in such amount, if any, as the Benefit Strategy Committee shall determine. Qualified Nonelective Non-ESOP Contributions made by an Employer shall be allocated to the Qualified Nonelective Non-ESOP Account of its employees who are both Participants and Non-Highly Compensated Employees. Qualified Nonelective ESOP Contributions made by an Employer shall be allocated to the Qualified Nonelective ESOP Account of its employees who are both Participants and Non-Highly Compensated Employees. Notwithstanding the foregoing, the Qualified Nonelective Contributions made under this Section on behalf of any individual who is a Participant and Non-Highly Compensated Employee shall not exceed the greater of 5% of such Participant's Compensation or two times the Plan's representative contribution rate.

4.06 <u>Rollover Contributions</u>. With the approval of the Plan Manager, a Participant (or the surviving Spouse of a deceased Participant who has an Account under the Plan) may contribute to a Rollover Account all or a portion of the amount payable to the Participant (or the surviving Spouse of a deceased Participant who has an Account under the Plan) as an eligible rollover distribution from an eligible retirement plan (as defined under section 401(a)(31) of the Code); provided, however, that (1) a former Participant may rollover to the Plan all or a portion of any eligible rollover distribution received from the Unisys Pension Plan or the Unisys Pension Plan 2, and (2) the Plan shall not accept a rollover from a Roth IRA described in section 408A of the Code but, effective January 1, 2020, shall accept a rollover of designated Roth contributions transferred directly from a designated Roth account under another employer's 401(k) plan or 403(b) plan. If the Plan receives such a rollover contribution Account. Any payment to the Plan pursuant to this Section 4.06 shall be made as a direct rollover that satisfies section 401(a)(31) of the Code or shall be made to the Plan within 60 days after the Participant's (or the surviving Spouse's) receipt of the distribution from the plan or individual retirement account in such manner as may be approved by the Plan Manager.

4.07 <u>Contribution Attributable to Military Service</u>. If a Participant returns to employment with the Employer following a period of service in the Armed Forces of the United States for which an Employer is required to give reemployment rights by law, the Employer contributions to the Plan with respect to such period shall be as follows:

(a) During the period that begins on the date of the Participant's return to employment and lasts for the lesser of (1) the product of 3 multiplied by the applicable period of military service; or (2) five years, the Participant may elect a Compensation

reduction in return for the corresponding Tax Deferred Contributions on his behalf, or After-Tax Contributions, as applicable, that could have been made if the Participant had continued to be employed and received Compensation during the applicable period of military service.

(b) The Employer shall contribute to the Plan, on behalf of each Participant who has been credited under subsection (a) with Tax Deferred Contributions or After-Tax Contributions, Matching Contributions equal to the amount of Matching Contribution that would have been required under Section 4.03 had such Tax Deferred or After-Tax Contributions, as applicable, been made during the applicable period of military service.

A Participant who is entitled to a contribution pursuant to this Section 4.07 shall not be entitled to receive corresponding retroactive earnings attributable to such contribution nor shall he be entitled to participate in the allocation of any forfeiture that occurred during his period of military service. For purposes of this Section 4.07, an Employee's Compensation for the applicable period of military service shall be deemed to equal the amount of Compensation the Employee would have received from the Employer during such period, based on the rate of pay the Employee would have received from the Employer but for the absence due to military service, or, if such rate of pay is not reasonably certain, the Employee's average Compensation during the 12-month period immediately before the qualified military service or, if shorter, the period of employment immediately before the qualified military service. The limitations under Sections 5.01 and 5.04 are applicable to contributions made pursuant to this Section 4.07 for the Plan Year to which the contributions relate. The limitations under Sections 5.02 and 5.03 shall not apply to contributions made pursuant to subsections (a) or (b) of this Section 4.07.

4.08 <u>Allocation of Payments Relating to Executive Life Insurance Company Insolvency</u>. To the extent the Plan is paid any amount from a state guaranty association with regard to the insolvency of Executive Life Insurance Company in 1991, such amount shall be allocated on a pro rata basis, in accordance with procedures adopted by the Plan Manager to the Accounts of any Participant who (a) resided in such state on the applicable trigger date for coverage under the state's guaranty association statute, and (b) had any portion of his Accounts invested, as of April 11, 1991, in a fund that held an Executive Life Insurance Company guaranteed investment contract. The specific Accounts to which a Participant's allocation shall be credited shall be the Accounts which were invested in the guaranteed investment contract.

4.09 <u>Form and Timing of Contributions</u>. Contributions shall be made to the Fund as soon as administratively practicable after the close of the payroll period to which they relate. In no event, however, shall Tax Deferred (including Roth) and After-Tax Contributions be made to the Fund later than the date prescribed under applicable regulations. In no event shall Matching Contributions be made to the Fund later than the last date on which amounts so paid may be deducted for federal income tax purposes by the contributing Employer for the taxable year in which the Plan Year ends. Generally, contributions shall be made in cash; provided, however, that Marching Contributions may be made in the form of Unisys Stock or cash, as determined by the Company in its sole discretion. The value of the Unisys Stock contributions is actually made to the Fund, determined in accordance with procedures established by the Plan Manager and the Trustee.

4.10 <u>Recovery of Employer Contributions</u>. The Employer may recover its contributions under the Plan as follows:

(a) if a contribution is made by an Employer under a mistake of fact, the excess of the amount contributed over the amount that would have been contributed had there not occurred a mistake of fact may be recovered by the Employer within one year after payment of the contribution; or

(b) if the contribution is conditioned upon its deductibility under section 404 of the Code, the contribution may be recovered, to the extent a deduction is disallowed, within one year after the disallowance.

Earnings attributable to an excess contribution may not be recovered by the Employer. Any losses attributable to the excess contribution shall reduce the amount the Employer may recover.

4.11 <u>Plan Expense Contributions</u>. The Employer, in its sole discretion, may contribute to the Plan, at any time and from time to time, such cash amounts as it shall determine in its sole discretion, which contributions shall be used to pay expenses of the Plan as determined by the Plan Manager. Such contributions shall be allocated as of the end of the Plan Year with respect to which such contribution is made, on a per capita basis, among all Participants who are employed on the last day of such Plan Year. Anything contained in this Article IV, Article VI, Article VI, Article X, or elsewhere in the Plan to the contrary notwithstanding, (i) Plan Expense Contributions may be made by the Employer for a Plan Year at any time, but not later than the date on which amounts so contributed may be deducted for federal income tax purposes by the contributing Employer for the taxable year on or within which such Plan Year ends; (ii) a Participant may not direct the investment of amounts credited to his Plan Expense Contribution Account, instead, such amounts shall be invested by the Investment Committee in short-term investments pending the use of such amounts to pay plan expenses; (iii) a Participant shall be fully vested in amounts credited to the Participant's Plan Expense Contribution Account; and (iv) no withdrawals or loans may be made by a Participant with respect to amounts credited to the Participant's Plan Expense Contribution Account.

4.12 <u>Roth In-Plan Conversions</u>. Subject to the requirements of this Section 4.12 and section 402A(c)(4) of the Code, and regulations and rulings promulgated thereunder, a Participant who is a Covered Employee may convert all or a portion of his vested Account (excluding amounts held in the Participant's Roth Contribution Account and amounts that are not currently available for a withdrawal or distribution) to Roth Contributions by making a Roth conversion election. Converted amounts shall be held in the Participant's Roth In-Plan Conversion Account. A Participant may make a Roth conversion election at any time following notification to the Committee, and in accordance with rules and procedures established by the Committee.

ARTICLE V

LIMITATIONS ON EMPLOYER CONTRIBUTIONS

5.01 Dollar Limitation on Tax Deferred Contributions (Including Roth Contributions).

(a) The Tax Deferred Contribution made on behalf of a Participant pursuant to Section 4.01(a) for a calendar year (including any portion of such Tax Deferred Contributions designated as Roth Contributions) shall not exceed the dollar limit specified under section 402(g) of the Code. This dollar limit shall be reduced by the amount, if any, contributed on behalf of the Participant under any other qualified cash or

deferred arrangement, simplified employee pension or annuity established under section 403(b) of the Code for the calendar year, other than elective deferral contributions made pursuant to section 414(v) of the Code.

(b) In the event that the dollar limit described in subsection (a) is exceeded in a calendar year for a Participant, the Plan Manager shall direct the Trustee to distribute by April 15 of the following calendar year, the amount of excess Tax Deferred Contributions, plus the sum of the allocable gain or loss on such excess Tax Deferred Contributions for the calendar year in which such Tax Deferred Contributions were made. If a Participant with excess Tax Deferred Contributions to the Plan during the applicable year, Tax Deferred Contributions (excluding Roth Contributions) not taken into account in determining Matching Contributions under Section 4.03 shall be reduced first, Roth Contributions not taken into account in determining Matching Contributions) taken into account in determining Matching Contributions) taken into account in determining Matching Matching Contributions under Section 4.03 shall be reduced second, Tax Deferred Contributions (excluding Roth Contributions) taken into account in determining Matching Contributions) taken into account in determining Matching Matching Contributions under Section 4.03 shall be reduced third, and Roth Contributions taken into account in determining Matching Contributions under Section 4.03 shall be reduced third, and Roth Contributions taken into account in determining Matching Contributions under Section 4.03 shall be reduced last.

(c) The Participant shall forfeit any Matching Contributions (excluding Matching Contributions forfeited or distributed pursuant to the provisions of Sections 5.03(b)(4) and (5)) and earnings, allocated to him or her by reason of the distributed Tax Deferred and Roth Contributions.

5.02 Limitation on Tax Deferred Contributions for Highly Compensated Employees.

(a) For each Plan Year the average of the Actual Deferral Percentages for Participants who are Highly Compensated Employees shall be compared to the average of the Actual Deferral Percentages for the other Participants for the current Plan Year; the average of the Actual Deferral Percentages for Participants who are Highly Compensated Employees shall not exceed the greater of:

(1) the average of the Actual Deferral Percentages for Participants who are Non-Highly Compensated Employees for the current Plan Year, multiplied by 1.25; or

(2) the lesser of:

(A) the average of the Actual Deferral Percentages for Participants who are Non-Highly Compensated Employees for the current Plan Year multiplied by two, or

(B) the average of the Actual Deferral Percentages for Participants who are Non-Highly Compensated Employees for the current Plan Year plus two.

In the event that the Plan satisfies the requirements of section 401(a)(4), 401(k) or 410(b) of the Code only if aggregated with one or more other qualified retirement plans, or if one or more other qualified retirement plans satisfy the requirements of these sections only if aggregated with the Plan, then this subsection (a) shall be applied as if all such plans were a single plan.

(b) If in the Plan Year, the average of the Actual Deferral Percentages for Participants who are Highly Compensated Employees exceeds the limit in subsection (a) for a Plan Year, the Plan Manager shall:

(1) determine the amount by which the Actual Deferral Percentage for Highly Compensated Employee or Employees with the highest Actual Deferral Percentage or Percentages for the Plan Year would need to be reduced to comply with the limit in subsection (a);

(2) convert the excess percentage amount determined under clause (1) into a dollar amount; and

(3) reduce the Tax Deferred Contributions of the Highly Compensated Employee with the greatest dollar amount of Tax Deferred Contributions made on their behalf with respect to the Plan Year pursuant to Section 4.01(a) by the lesser of (A) the amount by which the dollar amount of the affected Highly Compensated Employee's Tax Deferred Contributions made pursuant to Section 4.01(a) exceeds the dollar amount of the Highly Compensated Employee with the next highest dollar amount of Tax Deferred Contributions made pursuant to Section 4.01(a) exceeds the dollar amount of the Highly Compensated Employee with the next highest dollar amount of Tax Deferred Contributions made pursuant to Section 4.01(a), or (B) the amount of the excess dollar amount determined under clause (2); and

(4) either:

(A) direct the Trustee to return the excess Tax Deferred Contributions, as adjusted in accordance with subsection (d), to the individuals from whose Accounts the excess Tax Deferred Contributions were obtained within two and one-half months following the close of the Plan Year, if administratively practicable, but in no event later than the close of the following Plan Year;

(B) recharacterize the Tax Deferred Contribution as an After-Tax Contribution, to the extent permitted by the applicable Treasury regulations, no later than two and one-half months following the close of the Plan Year; or

(C) make Qualified Nonelective Non-ESOP Contributions, as described under Section 4.05, to the extent necessary to satisfy subsection (a).

(c) To the extent that a Matching Contribution relates to excess Tax Deferred Contributions returned or recharacterized pursuant to subsection (b)(4), such Matching Contributions, as adjusted in accordance with subsection (d), shall be forfeited immediately. Amounts forfeited during the Plan Year shall be used in accordance with the provisions of Section 7.02(d).

(d) The excess Tax Deferred Contributions returned or recharacterized pursuant to subsection (b), and any Matching Contributions forfeited pursuant to subsection (c) shall be adjusted for any income or loss thereon up to the date of distribution or forfeiture, as applicable, using the Plan's method for allocating income and loss as provided under Section 5.05.

(e) The amount of the excess Tax Deferred Contributions to be returned pursuant to subsection (b) for a Plan Year shall be reduced by the amount of excess Tax Deferred Contributions previously distributed to the Highly Compensated Employee pursuant to Section 5.01(b) for such Employee's taxable year ending on or within the Plan Year for which the excess Tax Deferred Contributions are returned pursuant to subsection (b).

(f) If a Participant with excess Tax Deferred Contributions under this Section has contributed Roth Contributions and other Tax Deferred Contributions to the Plan during the applicable year, Tax Deferred Contributions (excluding Roth Contributions) not taken into account in determining Matching Contributions under Section 4.03 shall be reduced first, Roth Contributions not taken into account in determining Matching Contributions under Section 4.03 shall be reduced second, Tax Deferred Contributions (excluding Roth Contributions) taken into account in determining Matching Contributions under Section 4.03 shall be reduced second, Tax Deferred Contributions (excluding Roth Contributions) taken into account in determining Matching Contributions under Section 4.03 shall be reduced third, and Roth Contributions taken into account in determining Matching Contributions under Section 4.03 shall be reduced last.

5.03 Limitation on After-Tax Contributions and Matching Contributions for Highly Compensated Employees.

(a) For each Plan Year the average of the Actual Contribution Percentages for Participants who are Highly Compensated Employees shall be compared to the average of the Actual Contribution Percentages for the other Participants for the current Plan Year; the average of the Actual Contribution Percentages for Participants who are Highly Compensated Employees shall not exceed the greater of:

(1) the average of the Actual Contribution Percentages for Participants who are Non-Highly Compensated Employees for the current Plan Year multiplied by 1.25; or

(2) the lesser of:

(A) the average of the Actual Contribution Percentages for Participants who are Non-Highly Compensated Employees for the current Plan Year multiplied by two, or

(B) the average of the Actual Contribution Percentages for Participants who are Non-Highly Compensated Employees for the current Plan Year plus two.

In the event that the Plan satisfies the requirements of section 401(a)(4), 401(m) or 410(b) of the Code only if aggregated with one or more other qualified retirement plans, or if one or more other qualified retirement plans satisfy the requirements of these sections only if aggregated with the Plan, then this subsection (a) shall be applied as if all such plans were a single plan.

(b) If in any Plan Year the average of the Actual Contribution Percentages for Participants who are Highly Compensated Employees exceeds the limit in subsection (a) for a Plan Year, the Plan Manager shall:

(1) determine the amount by which the Actual Contribution Percentage for Highly Compensated Employees with the highest Actual Contribution Percentage or Percentages for the Plan Year would need to be reduced to comply with the limit in subsection (a);

(2) convert the excess percentage amount determined under clause (1) into a dollar amount; and

(3) reduce the After-Tax Contributions (including any Tax Deferred Contributions recharacterized as After-Tax Contributions pursuant to Section 5.02(b)(4)(B)) and then, to the extent necessary, the Matching Contributions of the

Highly Compensated Employee with the greatest dollar amount of aggregate After-Tax and Matching Contributions made on their behalf with respect to the Plan Year by the lesser of (A) the amount by which the dollar amount of the affected Highly Compensated Employee's aggregate After-Tax and Matching Contributions exceeds the dollar amount of the Highly Compensated Employee with the next highest dollar amount of After-Tax and Matching Contributions, or (B) the amount equal to the excess dollar amount determined under clause (2); and

(4) either:

(A) direct the Trustee to return the excess After-Tax Contributions and vested Matching Contributions, as adjusted in accordance with subsection (c), to the individuals from whose Accounts the excess Matching Contributions were obtained within two and one-half months following the close of the Plan Year, if administratively practicable, but in no event later than the close of the following Plan Year; or

(B) make Qualified Nonelective Non-ESOP Contributions, as described under Section 4.05, to the extent necessary to satisfy the limit under subsection (a); and

(5) direct the Trustee to forfeit the excess unvested Matching Contributions, as adjusted in accordance with subsection (c), to the individuals from whose Accounts the excess Matching Contributions were obtained. Amounts forfeited during the Plan Year shall be used in accordance with the provisions of Section 7.02(d).

(c) To the extent that a Matching Contribution relates to excess After-Tax Contributions returned pursuant to subsection (b)(4), such Matching Contributions, as adjusted in accordance with subsection (d), shall be forfeited immediately. Amounts forfeited during the Plan Year shall be used in accordance with the provisions of Section 7.02(d).

(d) The excess After-Tax and Matching Contributions returned or recharacterized pursuant to subsection (b) shall be adjusted for any income or loss thereon up to the date of the distribution or forfeiture, as applicable, using the Plan's method for allocating income and loss as provided under Section 5.05.

- 5.04 Limitations on Allocations.
- (a) The maximum allowable addition to any Participant's Accounts for any Plan Year shall be the lesser of:
 - (1) \$40,000 (as adjusted under section 415(d) of the Code); or
 - (2) 100% of the Participant's Testing Compensation for the Plan Year.

For purposes of this Section 5.04, an addition shall not include Tax Deferred Contributions made pursuant to Section 4.01(b) and Rollover Contributions but shall include all other contributions and forfeitures allocated to a Participant's Accounts for the Plan Year, and all contributions and forfeitures under any other defined contribution plan of the Company or an Affiliate (other than elective deferral contributions made pursuant to section 414(v) of the Code).

(b) If the addition to any Participant's Accounts (other than his Tax Deferred Contributions made pursuant to Section 4.01(b) and his Rollover Account) for any Plan Year exceeds the maximum annual allowable addition to such Participant's Accounts under subsection (a), excess amounts shall be corrected as permissible under applicable guidance, including the Employee Plans Compliance Resolution System that is issued by the Internal Revenue Service.

5.05 <u>Distribution or Forfeiture of Income</u>. Any distribution or forfeiture of Tax Deferred Contributions, After-Tax Contributions or Matching Contributions necessary pursuant to Section 5.02 and 5.03 shall include a distribution or forfeiture of the income, if any, allocated to such contributions determined as of the last day of the Plan Year in which such contributions were made.

5.06 <u>Overall Deductibility Limit</u>. In no event may the aggregate contribution made by an Employer under the Plan for a Plan Year exceed the amount that may be deducted under section 404 of the Code with respect to such Plan Year.

ARTICLE VI

INVESTMENT AND VALUATION OF ACCOUNTS

6.01 Investment Direction by Participants. Except as otherwise provided in Section 6.02, each Participant shall direct the Trustee to invest the amounts credited to his Accounts in one or more Investment Funds, subject to the rules and procedures established by the Plan Manager. A Participant's investment direction shall be made at the time and in the manner prescribed by the Plan Manager. If any balance remains in a Participant's Accounts after his death, his Beneficiary shall direct the investment of the amounts credited to the Accounts as if the Beneficiary were the Participant. To the extent required by a Qualified Domestic Relations Order, the alternate payee of a Participant shall direct the investment of the amounts credited to the Participant's Accounts as though the alternate payee were the Participant. To the extent a Participant, Beneficiary or alternate payee directs the investment of the amounts credited to be subject to section 404(c) of ERISA, as described under Section 6.07. To the extent that a Participant, Beneficiary or alternate payee does not direct the investment of his Account, his or her Account shall be invested pending such direction in the Qualified Default Investment Alternative. Notwithstanding the foregoing, the Investment Committee shall have the right to adopt rules and procedures to govern Participant, Beneficiary or alternate payee investment elections under the terms of the Plan, whether or not such rules and procedures are required by the investment funds.

6.02 <u>Investment Funds</u>. The Investment Funds available under the Plan (other than the Unisys Common Stock Fund) shall be designated by, and at the sole discretion of, the Investment Committee, provided that (a) in no event shall there be more than 25 Investment Funds available under the Plan, including the Unisys Common Stock Fund as provided in Section 6.05, and one or more life-cycle or target-retirement-date funds whose assets are allocated based on each such fund's target date and (b) for any period during which the Plan is subject to section 401(a)(35) of the Code, in no event shall there be less than three Investment Funds available under the Plan (not including the Unisys Common Stock Fund). The Investment Committee, at its sole discretion, may from time to time designate or establish new investment funds or eliminate existing Investment Funds (other than the Unisys Common Stock Fund). Investment in any Investment Fund shall be made in accordance with rules formulated by the Investment Committee and the accounting procedures applied under the Plan shall be modified by

the Investment Committee to the extent they deem appropriate to reflect investments in that Investment Fund. The Investment Committee has the authority to select and appoint Investment Managers. The Investment Funds may be managed by the Trustee or an Investment Manager. Pending investment, reinvestment or distribution, as provided in the Plan, the Trustee or Investment Manager may temporarily retain the assets of any one or more Investment Funds in cash, commercial paper, short-term government obligations or, unless otherwise directed by the Investment Committee, undivided interests or participations in common or collective funds consisting of short-term investments, including funds of the Trustee or Investment Manager.

6.03 <u>Valuation of the Fund</u>. As of each Valuation Date, any increase or decrease in the fair market value of each Investment Fund (net after deduction of liabilities) since the preceding Valuation Date shall be credited to or deducted from the Accounts, if any, of each Participant. The allocation for each Investment Fund shall be made in the proportion that the balance in each Account invested in the Investment Fund as of the Valuation Date bears to the aggregate balance in all Accounts invested in the Investment Fund on that date. For purposes of the preceding sentence, the Employer's contributions to the Plan for the current year shall be excluded. The fair market value of investments shall be determined in accordance with any reasonable method permitted under regulations prescribed by the United States Department of the Treasury and such reasonable and uniform rules as the Trustee may adopt.

6.04 <u>Unisys Common Stock Fund</u>. The Investment Funds under the Plan shall include the Unisys Common Stock Fund, which is an Investment Fund providing for investment and reinvestment exclusively in Unisys Stock, except to the extent cash is held to facilitate purchases and sales within the fund. Investments in the Unisys Common Stock Fund shall be accounted for on the basis of units of the Unisys Common Stock Fund. Shares of Unisys Stock and cash received by the Unisys Common Stock Fund that are attributable to dividends, stock dividends, stock splits or to any reorganization or recapitalization of Unisys Corporation shall remain in or be invested in, as applicable, the Unisys Common Stock Fund and allocated to the Participant Accounts in proportion to the number of units of the Unisys Common Stock Fund held in such accounts. The transfer taxes, brokerage fees and other expenses incurred in connection with the purchase, sale or distribution of Unisys Stock, including Unisys Common Stock Fund shall be paid by the Unisys Common Stock Fund. In addition, the Unisys Common Stock Fund shall bear any other administrative fees and expenses incurred by the Plan in connection with the transfer of the Participant's interest in the Unisys Common Stock Fund. The voting and tendering of Unisys Stock held in the Unisys Common Stock Fund shall be subject to the following:

(a) For purposes of this Section, shares of Unisys Stock shall be deemed to be allocated and credited to each applicable Account of the Participant in an amount to be determined based on the balance in such account on the accounting date coincident with or next preceding the record date of any vote or tender offer and the closing price of Unisys Stock on such accounting date or if not traded on that date, on the business day on which shares of Unisys Stock were last traded before that accounting date.

(b) Each Participant who has any amounts under his Account invested in the Unisys Common Stock Fund shall be given notice by the Trustee of the date and purpose of each meeting of the stockholders of the Company at which shares of Unisys Stock are entitled to be voted, and instructions shall be requested from each such Participant as to the voting at the meeting of such Unisys Stock. If the Participant furnishes instructions within the time specified in the notification given to him, the

Trustee shall vote such Unisys Stock in accordance with the Participant's instructions. Shares of Unisys Stock that have not been credited to any Participant's Account or for which no instructions were timely received by the Trustees, whether or not credited to the Account of any Participant shall be voted by the Trustee in the same proportion that the allocated and voted shares of Unisys Stock have been voted by Participants. The Investment Committee shall establish procedures under which notices shall be furnished to Participants as required by this subsection (b) and under which the Participants' instructions shall be furnished to the Trustee.

(c) Each Participant who has any amounts under his Account invested in the Unisys Common Stock Fund shall be given notice of any tender offer for, or a request or invitation for tenders of, Unisys Stock made to the Trustees. Instructions shall be requested from each such Participant as to the tendering of shares of Unisys Stock credited to his Account and for this purpose Participants shall be provided with a reasonable period of time in which they may consider any such tender offer for, or request or invitation for tenders of, Unisys Stock made to the Trustees. The Trustees shall tender such Unisys Stock as to which the Trustees have received instructions to tender from Participants within the time specified. Unisys Stock credited to an Account as to which the Trustee has not received instructions from a Participant shall not be tendered. Shares of stock that have not been credited to any Participant's Account shall be tendered by the Trustee in the same proportion that the allocated and tendered shares of Unisys Stock have been tendered by Participants. The Investment Committee shall establish procedures under which notices shall be furnished to the Trustee. In carrying out their responsibilities under this subsection (c) the Trustees may rely on information furnished to them by (or under procedures established by) the Investment Committee.

(d) For all purposes of this Section 6.05, the number of shares of Unisys Stock held in a Participant's Account which are invested in the Unisys Common Stock Fund shall be the number of shares of Unisys Stock represented by the number of units held in such accounts after reducing such number of units by the number of units in such accounts which represent cash.

(e) With respect to Participants subject to Section 16 of the Securities Exchange Act of 1934, the Investment Committee shall apply any requirements or restrictions required for the Plan to obtain the protections of Rule 16b-3 under the Securities Exchange Act of 1934 or any successor Rule or regulation intended to replace Rule 16b-3.

6.05 <u>Special Rule Regarding Appraisal of Unisys Stock</u>. If at any time the Unisys Stock held by the ESOP Portion of the Plan is not readily tradable on an established securities market, all valuations of such Unisys Stock with respect to activities carried on by the Plan shall be made by an independent appraiser meeting requirements similar to the requirements of Treasury Regulation §1.170A-13(c)(5).

6.06 <u>Section 404(c) Compliance</u>. The Plan is intended to constitute a plan described in section 404(c) of ERISA and section 2550.404c-1 of the United States Department of Labor regulations. Thus, no fiduciary of the Plan shall be liable for any loss, or by reason of any breach, which results from any investment direction made by a Participant, Beneficiary or alternate payee under a Qualified Domestic Relations Order. The Company or its delegate shall comply with, or monitor compliance with, as required, all disclosure and other responsibilities described in sections 2550.404c-1(b)(2)(i)(A) and (b)(2)(i)(B)(1) of the United States Department of Labor regulations except that the

Trustee shall monitor compliance with those procedures established to provide confidentiality of information relating to the exercise of voting and tender rights by Participants. If the Company determines that a situation has potential for undue influence by the Company, the Company shall direct an independent party to perform such activities as are necessary to ensure the confidentiality of the rights of Participants.

ARTICLE VII

VESTING

7.01 Vesting Schedule.

(a) A Participant shall at all times be fully vested in the balance of his After-Tax Account, Tax Deferred Account, Roth Contribution Account, Roth In-Plan Conversion Account, GPEP Account, Tax Deductible Contribution Account, and Rollover Account.

(b) A Participant employed by an Employer on or after January 1, 2000 shall be fully vested in his ESOP Account and Regular Account. A Participant who terminated employment with all Employers prior to January 1, 2000 shall have his vested interest in his ESOP Account and Regular Account determined in accordance with the provisions of the Plan as in effect on the date he terminated employment with all Employers.

(c) A Participant's vested interest in his Matching Contribution Account shall be determined under the following schedule:

Years of Service	Vested Percentage
Less than 1	0%
At least 1 year, but less than 2 years	33%
At least 2 years, but less than 3 years	66%
3 years or more	100%

Notwithstanding the foregoing, a Participant shall be fully vested in his Matching Contribution Account if (1) he attains Normal Retirement Age while employed by an Employer or an Affiliate or (2) he was hired prior to May 1, 2022.

(d) If a Participant receives a distribution from his Matching Contributions Account, at a time when he is not fully vested in such Account, his vested interest in such Account at all times prior to the date on which he incurs his fifth consecutive Period of Severance shall be determined by the formula:

X = P (AB + D) - D

where X is the vested interest in the applicable Account, P is the percentage determined in accordance with this Section at the time of determination, AB is the balance of the applicable Account at the time of determination and D is the amount which was previously distributed to the Participant from the applicable Account.

7.02 Forfeitures.

(a) The unvested portion of a Participant's Accounts shall be forfeited as of the earlier of the date described in paragraphs (1) and (2) below:

(1) as of the last day of the Plan Year in which a Participant incurs a Period of Severance equal to five consecutive years;

(2) the last day of the Plan Year in which the Participant receives a distribution of his vested interest under the Plan.

(b) For purposes of subsection (a), a Participant who terminates employment with the Employer and all Affiliates and has no vested interest in his Accounts at such time, shall be deemed to have received a single sum payment of his entire vested interest in his Accounts as of the date of his Termination of Employment. Restorations pursuant to this subsection (b) shall be made from currently forfeited accounts in accordance with subsection (d), or from additional contributions by the Employer.

(c) If a Participant who has received a distribution described in paragraph (a)(2) of this Section, whereby any part of his Account has been forfeited, again becomes a Covered Employee prior to incurring five (5) consecutive twelve-month Periods of Severance and repays the amount of the distribution no later than the fifth anniversary of the date on which the Participant again becomes a Covered Employee, the amount so forfeited shall be restored to his Account.

(d) Amounts forfeited under the Plan shall be used to restore amounts required to be restored in accordance with subsection (c), reduce Employer Matching Contribution, or pay Plan expenses.

ARTICLE VIII

AMOUNT OF BENEFITS

8.01 <u>Benefits Upon Severance from Employment</u>. A Participant who incurs a Severance from Employment for a reason other than death shall be entitled to a distribution of the entire vested balance of his Accounts as of the Valuation Date coincident with or immediately preceding his Benefit Commencement Date.

8.02 <u>Death Benefits</u>. If a Participant's Severance from Employment occurs by reason of his death, his Beneficiary shall be entitled to a distribution of the entire vested amount credited to the Participant's Accounts as of the Valuation Date coincident with or next following his Benefit Commencement Date.

ARTICLE IX

PAYMENT AND FORM OF BENEFITS

9.01 Form of Benefit Paid to Participant.

(a) Unless a Participant elects otherwise in accordance with subsection (b), any benefit due a Participant under Article IX shall be paid in a single sum, subject to 9.04. If the vested Account balance to which a Participant is entitled is zero as of the date of the Participant's Severance from Employment, such Participant shall be deemed

to have received a single sum payment of his entire vested Account balance under the Plan as of such date.

(b) If a Participant's vested Account balance exceeds \$1,000 as of his Benefit Commencement Date, he may, in lieu of the single sum payment prescribed under subsection (a), elect (1) monthly, quarterly, semi-annual or annual installments payable over a period of no less than one-year and no greater than 20 years; provided, however, that a Participant may, at any time, take a lump sum distribution of his remaining Account, or otherwise change or discontinue the installment payments or (2) a partial withdrawal of his Account at such times and in such amounts as may be requested by the Participant; provided that such election must be in writing and be made within the Notice Period in the manner prescribed by the Plan Manager. The Participant shall be provided with information regarding the consequences of failing to defer distribution of his vested Account balance until such later date as permitted under the Plan.

9.02 Benefit Commencement Date.

(a) Except as provided under this Article IX, if the Participant's vested Account balance as of his Benefit Commencement Date does not exceed \$1,000, his benefit under the Plan shall be paid in a single sum as soon as administratively practicable following the Valuation Date coinciding with or next following date of the Participant's termination of employment with Employer.

(b) Except as otherwise provided under this Article IX, if the Participant's vested Account balance as of his Termination of Employment is greater than \$1,000, the benefit payable to a Participant in accordance with Article VIII shall be paid or commence no later than the April 1 of the calendar year following the later of (1) the calendar year in which the Participant attains age 72 (or age 70½ with respect to a Participant who attained age 70½ on or before December 31, 2019) or (2) the calendar year in which the Participant incurs a Termination of Employment; provided, however, that the Participant may elect, in writing, to have his benefit paid or commence on the first day of any earlier month beginning with the month following the month in which his Termination of Employment occurred.

9.03 Form and Payment of Death Benefit.

(a) A Participant shall designate a Beneficiary or Beneficiaries to receive any benefits which may be payable under the Plan in the event of his death.

(b) If the vested Account balance to which a Beneficiary is entitled is \$1,000 or less, such amount shall be paid in a single sum, subject to Section 9.04. If the Account balance payable upon a Participant's death is zero, the Participant's Beneficiary shall be deemed to have received a single sum payment of the Participant's entire Account balance under the Plan or on the date of the Participant's death.

(c) If the vested Account balance exceeds \$1,000, the Participant's vested Account balance shall be paid to his Beneficiary in a single sum, subject to Section 9.04; provided, however, that if a Participant dies before his Benefit Commencement Date, his Beneficiary may elect, in lieu of a single sum payment, monthly installment payments over a period of no less than the life expectancy of the Beneficiary.

(d) Subject to Section 9.03(e), if a Participant dies on or after his Benefit Commencement Date but before the entire amount of his benefit has been paid, the

remaining amount shall be paid to his Beneficiary in the form and over the period being used at the Participant's date of death.

(e) <u>Distributions Upon the Death of a Participant under the Setting Every Community Up for Retirement</u> <u>Enhancement (SECURE Act)</u>. Notwithstanding the foregoing provisions of this Section, distributions attributable to a Participant who dies after 2019 shall be subject to the following provisions:

(1) the only distributions payable upon the Participant's death that are permitted to be paid over the life expectancy of a Designated Beneficiary (as defined below) are those distributions that are payable to an Eligible Designated Beneficiary (as defined below). Any post-death distributions that are payable to any Designated Beneficiary other than an Eligible Designated Beneficiary must be paid in full within 10 years of the Participant's death.

(2) For purposes of these post-death distribution requirements, a Designated Beneficiary is the individual who is designated as the Participant's beneficiary under Section 2.10 of the Plan and is the designated beneficiary under Section 401(a)(9) of the Code and Treas. Reg. \$1.401(a)(9)-1, Q&A-4.

(3) For purposes of these post-death distribution requirements, an Eligible Designated Beneficiary is:

- (A) The Participant's surviving Spouse.
- (B) The Participant's child who has not reached the age of majority.
- (C) A disabled person, as defined in section 72(m)(7) of the Code.

(D) A person certified to be chronically ill (as defined in section 7702B(c)(2) of the Code), for a reasonably lengthy period or indefinitely.

(E) A person who is not more than 10 years younger than the Participant.

(4) For purposes of any post-death distributions payable to an Eligible Designated Beneficiary:

(A) The post-death distribution shall begin within one year of the Participant's death (or such other date as may be permitted under section 401(a)(9) of the Code and regulations issued thereunder), except that in the case of benefits payable to the Participant's surviving Spouse, the post-death distribution shall begin no later than what would have been the Participant's required beginning date (or such other date as may be permitted under section 401(a)(9) of the code and regulations issued there as may be permitted under the participant's required beginning date (or such other date as may be permitted under section 401(a)(9) of the Code and regulations issued thereunder).

(B) A minor child will cease to be an Eligible Designated Beneficiary when he or she reaches the age of majority, at which time any remaining balance in the Participant's account balance must be fully distributed to such child within 10 years of the minor child reaching the age of majority.

(C) If an Eligible Designated Beneficiary dies, any remaining balance in the Participant's account balance must be fully distributed within 10 years.

(5) A Designated Beneficiary, who is not an Eligible Designated Beneficiary, must receive his or her distribution within 10 years of the Participant's death, except that if the Participant died before January 1, 2020, distributions can continue to be made over the Designated Beneficiary's lifetime. Upon the death of the Designated Beneficiary, any remaining balance in the Participant's account balance must be fully distributed within 10 years of the Designated Beneficiary's death.

(6) A Beneficiary who is not a Designated Beneficiary or an Eligible Designated Beneficiary must receive his or her distribution within 5 years of the Participant's death, except that if the Participant died before January 1, 2020, distributions can continue to be made over the beneficiary's lifetime.

9.04 <u>Form of Single Sum Distributions</u>. If a benefit under the Plan is payable in a single sum, such amount shall generally be paid in cash. However, a Participant or Beneficiary entitled to a distribution may elect, in the form and manner prescribed by the Plan Manager, to receive the vested balance of the Account invested in the Unisys Common Stock Fund in the form of whole shares of Unisys Stock (and cash with respect to fractional shares). Before any distribution is made from the Plan in a single sum, the portion of a Participant's ESOP Account that has been invested in Investment Funds other than the Unisys Common Stock Fund, shall be automatically reinvested in the Unisys Common Stock Fund before distribution.

9.05 <u>Put Options</u>. If the Unisys Stock held under the ESOP Portion of the Plan is not readily tradable on an established securities market (within the meaning of section 409(h)(1)(B) of the Code), any Participant who is entitled to a distribution of such shares from the Plan shall have a right to require the Company to repurchase such shares during the 60-day period following the date such shares are distributed or during an additional period of at least 60 days during the following Plan Year. Amounts paid by the Company under this Section shall be paid in substantially equal periodic payments (not less frequently than annually) over a period beginning not later than 30 days after the exercise of the put option described in the preceding sentence and not exceeding five years and adequate security shall be provided, and interest paid, on any unpaid installments. Except as otherwise provided herein or required by law, Unisys Stock held under the ESOP Portion of the Plan shall not be subject to a put, call, or other option, or a buy-sell or similar arrangement either while held by the Plan or when distributed to or on account of a Participant whether or not the Plan is then an Employee Stock Ownership Plan.

9.06 <u>Direct Rollovers</u>. In the event any payment or payments to be made under the Plan to a Participant, a Beneficiary who is the surviving Spouse of a Participant, or an alternate payee who is the former spouse of a Participant, would constitute an "eligible rollover distribution," such individual may request that such payment or payments be transferred directly from the Plan to the trustee of an "eligible retirement plan." Any such request shall be made in writing, on the form prescribed by the Plan Manager for such purpose, at such time in advance as the Plan Manager may specify.

For purposes of Section 9.06, an "eligible rollover distribution" shall mean a distribution from the Plan, excluding (1) any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) over the life (or life expectancy) of the individual, the joint lives (or joint life expectancies) of the individual and the

individual's designated Beneficiary, or a specified period of ten or more years, (2) any distribution to the extent such distribution is required under section 401(a)(9) of the Code, (3) any hardship distribution described in section 401(k) (2)(B)(i)(IV) of the Code; and (4) any other distribution that does not qualify as eligible for rollover. A portion of a distribution shall not fail to be an eligible rollover distribution merely because the portion consists of After-Tax Contributions which are not includible in gross income. The nontaxable portion of an "eligible rollover distribution" may be rolled over tax-free to an eligible rollover plan as specified below if the eligible rollover plan provides for separate accounting of the amount transferred and earnings on such amounts.

For purposes of Section 9.06, an "eligible retirement plan" shall mean (i) an individual retirement account described in section 408(a) of the Code, (ii) an individual retirement annuity described in section 408(b) of the Code (other than an endowment contract), (iii) an annuity plan described in section 403(a) of the Code, (iv) a qualified plan under section 401(a) of the Code, the terms of which permit the acceptance of rollover distributions, (v) an eligible deferred compensation plan described in section 457(b) of the Code that is maintained by an eligible employer described in section 457(e)(i)(A) of the Code that shall separately account for the distribution, or (vi) an annuity contract described in section 403(b) of the Code; provided, however, that (A) effective January 1, 2007, with respect to a distribution (or portion of a distribution) consisting of After-Tax Contributions, 'eligible rollover plan' shall mean a plan described in clause (i), (ii), (iii), (iv) or (vi), and (B) effective January 1, 2020, an eligible rollover plan for the purpose of a direct rollover of a distribution from a Participant's Roth Contribution Account and/or Roth In-Plan Conversion Account shall include only another designated Roth account under an applicable retirement plan described in Code Section 402A(e)(1) or a Roth IRA described in Code Section 408A.

A "qualified rollover contribution" as described in section 408A(e) of the Code may be made from the Plan to a Roth individual retirement account in a direct rollover subject to the rules set forth in section 408A of the Code and any regulations issued there under.

Any distribution of benefits to the Beneficiary of a deceased Participant who is not the surviving Spouse of the Participant may be transferred in a direct transfer to an individual retirement account or annuity under sections 408(a) and (b) of the Code established for the purpose of receiving such distribution and which will be treated as an inherited individual retirement account pursuant to the provisions of section 402(c)(11) of the Code, if such distribution otherwise meets the requirements set forth above. Such direct rollover of a distribution by a nonspouse Beneficiary shall be treated as an eligible rollover distribution only for purposes of section 402(c) of the Code. An eligible retirement plan shall include an individual retirement account or annuity under sections 408(a) and (b) of the Code established for the purpose of receiving a distribution that is rolled over from a nonspouse distributee, but only if the conditions set forth above are satisfied. Distributee shall include a nonspouse Beneficiary, but only if the conditions set forth above are satisfied.

9.07 <u>Minimum Required Distribution</u>. Notwithstanding anything to the contrary herein:

(a) a Participant who is a five percent owner of the Employer (as determined under section 416 of the Code) must begin to receive minimum distributions under the Plan pursuant to Section 401(a)(9) of the Code and related regulations no later than the April 1 following the calendar year in which he attains age 72 (or age 70½ with respect to a Participant who attained age 70½ on or before December 31, 2019). Any other Participant must begin to receive such minimum distributions upon the later of the April

1 following the calendar year in which he attains age 72 (or age 70½ with respect to a Participant who attained age 70½ on or before December 31, 2019) or incurs a Termination of Employment.

(b) the amount of minimum required distributions shall be determined and made in accordance with section 401(a)(9) of the Code, including the incidental death benefit rules of section 401(a)(9)(G) of the Code, and final regulations issued thereunder, including Treas. Reg. §1.401(a)(9)-2 through 1.401(a)(9)-9.

(c) any distribution otherwise required by this Section for 2020 shall be made only at the election of the Participant or Beneficiary and certain waived required distributions paid to a Participant in 2020 or 2021 may be eligible for rollover in accordance with the Plan's administrative procedures and IRS Notice 2020-51.

ARTICLE X

WITHDRAWALS AND LOANS

10.01 <u>General</u>. A Participant may withdraw amounts from his Account to the extent provided under this Article X and, if applicable, in accordance with Appendix B. Any withdrawal shall be considered the distribution of a portion of the Participant's benefit and shall be paid in a single sum. A withdrawal shall be disregarded, however, for purposes of determining whether the Participant's Benefit Commencement Date has occurred. A Participant's request for a withdrawal must be made in writing within the period prescribed by the Plan Manager. The amount of the withdrawal shall be divided proportionally among the Investment Funds in which the Accounts from which the withdrawal is to be made are invested. Withdrawals shall be made in accordance with the procedures established by the Plan Manager.

10.02 <u>Withdrawals from After-Tax Account</u>. Subject to the requirements set forth in Section 10.01, a Participant who is an Employee may withdraw all or a portion of the balance of his After-Tax Account (other than earnings on After-Tax Contributions made on or after January 1, 1987), up to one time in any six-consecutive month period. Withdrawals from a Participant's After-Tax Account shall be made in the following order:

(a) After-Tax Contributions made before January 1, 1987; then

(b) Amounts relating to After-Tax Contributions after December 31, 1986, including a pro-rata portion of the earnings thereon; and then

(c) Earnings on After-Tax Contributions made before January 1, 1987.

10.03 <u>Withdrawals from Tax Deductible Contribution Account and Rollover Account</u>. Subject to the requirements set forth in Section 10.01, a Participant may withdraw all or a portion of the balance of his Tax Deductible Contribution Account or Rollover Account at any time.

10.04 <u>Withdrawals from Regular Account</u>. Subject to the requirements set forth in Section 10.01, a Participant who is an Employee may withdraw all or a portion of the balance of his Regular Account, up to one time in any six-consecutive month period if the following requirements are met:

- (a) the Participant has withdrawn the entire balance of his After-Tax Account; and
- (b) the Participant's aggregate years of participation in this Plan and any Prior Plan is five years.

10.05 <u>Withdrawals from ESOP Account</u>. Subject to the requirements set forth in Section 10.01, a Participant who is an Employee may withdraw all or a portion of the vested balance of his ESOP Account (other than the portion of his ESOP Account attributable to Matching Contributions made on or after January 1, 2007), up to one time in any six-consecutive month period if the following requirements are met:

- (a) the Participant has withdrawn the entire balance of his After-Tax Account and his Regular Account; and
- (b) the Participant's aggregate years of participation in this Plan and any Prior Plan is five years.

10.06 <u>Withdrawals from GPEP Account</u>. Subject to the requirements set forth in Section 10.01, a Participant who is an Employee and who has withdrawn the entire balance of his After-Tax Account and his Regular Account may, up to one time in any six consecutive month period, withdraw the portion of the balance of his GPEP Account attributable to Contributions made at least 36-months prior to the date the withdrawal is requested.

10.07 Hardship Withdrawals.

(a) Subject to the requirements set forth in Section 10.01 and, if applicable, in subsection (b) of this Section 10.07, a Participant may elect a withdrawal from his Tax Deferred Account on account of an immediate and heavy financial hardship; provided, however, that (1) for periods prior to January 1, 2019, a Participant shall not be permitted to withdraw earnings credited to the Participant's Salary Deferral Account after December 31, 1988 and (2) the amount of such withdrawal is necessary to satisfy the immediate and heavy financial need as determined below.

(b) In the event a Participant receives a withdrawal under this Section 10.07, the Participant shall be both ineligible to have Tax Deferred Contributions made on his behalf and ineligible to make After-Tax Contribution for a period ending on the six (6) month anniversary of the date the Participant received the withdrawal; provided, however, that the suspension of contributions described in this subsection shall not apply after December 31, 2018.

(c) For purposes of this Section 10.07, an immediate financial hardship is expenses incurred as a result of:

(1) medical care described in section 213(d) of the Code incurred by the Participant, the Participant's spouse, any dependents of the Participant as defined in Treas. Reg. Section 1.401(k)-1(d)(3)(iii)(B)(3) or the Participant's primary Beneficiary (or the distribution is necessary for such persons to obtain such medical care);

(2) the purchase (excluding mortgage payments) of a principal residence for the Participant;

(3) the payment of tuition and related educational fees for the next twelve (12) months of postsecondary education for the Participant, his spouse, children, dependents (as defined in section 152 of the Code, without regard to section 152(b)(1), (b)(2) and (d)(1)(B) of the Code), or primary Beneficiary;

(4) the repair of damage to the Participant's principal residence that would qualify for the casualty deduction under section 165 of the Code (determined without regard to whether the loss exceeds 10% of adjusted gross income or, effective January 1, 2018, without regard to whether the loss is attributable to a federally declared disaster);

(5) the need to prevent the eviction of the Participant from, or foreclosure on the mortgage of, the Participant's principal residence;

(6) payments for burial or funeral expenses for the Participant's deceased parent, spouse, children, or dependents (as defined in section 152 of the Code, without regard to section 152(d)(1)(B) of the Code), or primary Beneficiary;

(7) federal, state or local income taxes or penalties reasonably anticipated to result from the distribution;

(8) effective January 1, 2019, expenses and losses incurred by the Participant on account of a disaster declared by the Federal Emergency Management Agency, provided the Participant's primary residence or place of employment is located in an area designated by the Federal Emergency Management Agency for individual assistance in response to the disaster; or

(9) such other circumstances as may be prescribed by the Secretary of the Treasury or his delegate.

The final determination of whether an immediate and heavy financial hardship exists shall be determined by the Plan Manager, which shall be under no obligation to verify independently the facts of hardship submitted by a Participant. Unless the Plan Manager or its designee has actual knowledge to the contrary, the Plan Manager shall be entitled to rely upon an affidavit signed by the Participant as proof of the elements necessary for a hardship withdrawal.

(d) For purposes of this Section 10.07, a withdrawal shall be deemed to be in the amount necessary to alleviate an immediate financial hardship if:

(1) for periods prior to January 1, 2019, (A) the amount of the withdrawal does not exceed the amount required to satisfy the immediate and heavy financial need; (B) the Participant has obtained all available withdrawals and distributions from his Regular Account, ESOP Account, GPEP Account, Tax Deductible Contribution Account, Rollover Account, and After-Tax Contribution Account; and (C) the Participant has obtained all nontaxable loans currently available to the Participant from the Plan and all plans maintained by the Company or an Affiliate; or

(2) for periods beginning on or after January 1, 2019, (A) the amount of the withdrawal does not exceed the amount required to satisfy the immediate and heavy financial need; (B) the Participant has obtained all currently permissible distributions (other than hardship distributions of elective deferrals) under this and all other qualified and nonqualified deferred compensation plans maintained by the

Employer or an Affiliate; and (C) the Participant represents in writing that he has insufficient liquid assets to satisfy the need.

10.08 <u>Withdrawals after Age 59½</u>. Subject to the requirements set forth in 10.01, after he has attained age 59½, a Participant may withdraw all or any portion of his vested interest in his Account, up to one time in any six-consecutive month period; provided, however, that the "one time in any six-consecutive month period" rule shall not apply to any Participant whose Account includes amounts transferred to the Plan from the Unify Plan.

10.09 <u>Military Withdrawals</u>. Effective January 1, 2009, a Participant receiving differential military pay shall be treated as having a Severance from Employment for purposes of taking a distribution of that portion of his or her Account consisting of Tax Deferred Contributions if he or she is absent from employment due to performing service in the uniformed services described in section 3401(h)(2)(A) of the Code. If a Participant elects to take a distribution pursuant to the foregoing, he or she shall be precluded from electing to have the Employer contribute Tax Deferred Contributions, Roth Contributions and After-Tax Contributions from his or her Compensation on his or her behalf to the Plan (and pre-tax and after-tax contributions under any other qualified or nonqualified plan of deferred compensation maintained by and Employer) for six months following the date of the distribution, provided, however, that the six-month suspension period shall not apply to a withdrawal taken under this Section after December 31, 2021 if the Participant has been ordered or called to active duty for a period in excess of 179 days or for an indefinite period.

10.10 <u>Loans to Participants</u>. The Plan Manager may, in his discretion, cause the Plan to lend to any qualified Participant an amount, as requested by the Participant, from his Accounts (excluding amounts held in his Tax Deductible Contribution Account or GPEP Account), upon such terms as the Plan Manager may see fit and, if applicable, in accordance with Appendix B.

(a) <u>Qualification for Loans</u>. A Participant is eligible for a Plan loan if he is (1) an Employee, or (2) a Participant who is a party in interest, as determined under section 3(14) of ERISA.

- (b) <u>Amount of Loan</u>. The amount lent to any Participant shall not exceed the lesser of:
 - (1) the lesser of \$50,000 or 50% of the amount in the Participant's vested interest in his Accounts; or
 - (2) the greater of \$10,000, or one-half of the value of the vested portion of the Employee's Account.

For purposes of determining the maximum amount of a loan under this subsection (b), the balance of a Participant's Tax Deductible Contribution Account and GPEP Account shall be disregarded. The minimum amount of any loan made to a Participant shall be set by the Plan Manager from time to time, in a uniform and nondiscriminatory manner. A Participant may not have more than one loan outstanding at any time.

(c) <u>Loan Term; Interest Rates</u>. Each loan shall be repaid within no less than one year and no more than five years from the date the loan is made, unless the loan proceeds are used to acquire a dwelling that is to be used as the Participant's principal residence, in which event the term of the loan may not be more than fifteen years. Each

loan shall bear a fixed rate of interest that is commercially reasonable, as determined by the Plan Manager.

(d) <u>Other Loan Requirements</u>. The amount lent to any Participant shall be debited against all of the Participant's Accounts from which the loan may be made (as determined under subsection (a)) such that the amount of the loan is prorated among such Accounts on the basis of the balance of each Account at the time the loan is made, and the interest paid to the Trustee by the Participant on the loan shall be allocated to such Accounts and to the Account of no other Participant. The amount of any loan, including accrued interest, un-repaid at the time a Participant or his Beneficiary becomes entitled to a distribution under Article IX shall be deducted from the amount otherwise distributable to the Participant or Beneficiary. No note or other document evidencing a loan shall be negotiable or otherwise assignable.

(e) <u>Elections</u>. In order to be valid, a Participant's request for a loan must be made in the time and manner prescribed by the Plan Manager.

(f) <u>Expense of Loan</u>. The Plan Manager may charge a reasonable loan processing fee as well as an annual loan administration fee for each year the loan is outstanding. Such fee shall be applied on a uniform and nondiscriminatory manner.

(g) <u>Repayment</u>. Loans shall be repaid in equal installments (not less frequently than quarterly) through payroll withholding or, in the case of a Participant who is on a unpaid leave of absence, layoff or has incurred a Termination of Employment, automatic monthly deductions from the Participant's bank account. A Participant may fully repay the loan at any time without penalty. Loans shall become immediately due and payable upon a Participant's death.

(h) <u>Loan Security and Documentation</u>. A loan shall be evidenced by a written document containing such terms and conditions as the Plan Manager shall determine, and shall be secured by the Participant's vested interest in his Accounts (other than his Tax Deductible Contributions Account).

(i) <u>Hurricane Loans.</u>

(1) Within the period permitted under applicable legislation, a Qualified Hurricane Katrina Participant, a Qualified Hurricane Rita Participant or a Qualified Hurricane Wilma Participant may obtain a loan from the Plan (after taking into account the outstanding balance of other loans) in an amount equal to the lesser of \$100,000 or 100 percent of the vested portion of the Participant's Account (less the highest value of all other outstanding loans in the prior 12 months).

(2) Any loan repayment otherwise due on or after (1) August 25, 2005 through December 31, 2006 in the case of a Qualified Hurricane Katrina Participant, (2) September 23, 2005 through December 31, 2006 in the case of a Qualified Hurricane Rita Participant or (3) October 23, 2005 through December 31, 2006 in the case of a Qualified Hurricane Wilma Participant shall be delayed for one year. After the one-year delay, such Participant's loan repayments shall be adjusted to reflect the delayed repayments and unpaid interest. The loan repayment term shall be extended by one year regardless of whether such extension would cause the loan original loan term to extend beyond five years in the case of loan not used to purchase a Participant's principal residence.

(3) For purposes of this Section:

(A) "Qualified Hurricane Katrina Participant" means an individual whose principal place of residence on August 28, 2005 was located in the Hurricane Katrina disaster area and who has sustained an economic loss by reason of Hurricane Katrina.

(B) "Qualified Hurricane Rita Participant" means an individual whose principal place of residence on September 23, 2005 was located in the Hurricane Rita disaster area and who has sustained an economic loss by reason of Hurricane Rita.

(C) "Qualified Hurricane Wilma Participant" means an individual whose principal place of residence on October 23, 2005 was located in the Hurricane Wilma disaster area and who has sustained an economic loss by reason of Hurricane Wilma.

(j) Qualified Military Service.

(1) Effective for loans granted on or after January 1, 2004, a Participant shall have his loan repayments suspended during an absence from employment with the Employer and its Affiliates due to qualified military service under section 414(u) of the Code and the term of the Participant's loan shall be increased by the length of the qualified military service.

(2) If a Participant has an outstanding loan under the Plan on the date he commences a leave of absence for qualified military service under section 414(u) of the Code, the interest rate on the Participant's loan shall not exceed 6% per year for the duration of the qualified military service.

(k) <u>Coronavirus Loan Deferment</u>. A Qualified Coronavirus Participant may defer, until January 1, 2021, any loan repayment otherwise due during the period specified by the Coronavirus Aid, Relief, and Economic Security Act of 2020. After the delay, such Participant's loan repayments shall be adjusted to reflect the delayed repayments and unpaid interest. The loan repayment term shall be extended by the loan deferral period regardless of whether such extension would cause the term of the loan to extend beyond the term described in Section 10.10(c) of the Plan.

For purposes of this Section, a Qualified Coronavirus Participant is a Participant who has certified, in accordance with procedures established by the Trustee, that he (1) has been diagnosed, or has a spouse or dependent who has been diagnosed, with SARS-CoV-2 or COVID-19 using a CDC-approved test, or (2) has experienced adverse financial consequences as a result of the Participant, the Participant's spouse, or a member of the Participant's household (A) being quarantined, being furloughed or laid off, or having work hours reduced due to SARS-CoV-2 or COVID-19, (B) being unable to work due to lack of childcare due to SARS-CoV-2 or COVID-19, (C) being affected by the closure or reduction in hours of a business that they own or operate due to SARS-CoV-2 or COVID-19, (D) having pay or self-employment income reduced due to SARS-CoV-2 or COVID-19, or (E) having a job offer rescinded, or start date for a job delayed, due to SARS-CoV-2 or COVID-19.

(I) Unify Plan Loans. Loans transferred to the Plan from the Unify Plan shall be administered in accordance with the applicable provisions of the Unify Plan and the loan agreement, including any loan repayment deferrals permitted under the Unify Plan in accordance with the applicable provisions of the Coronavirus Aid, Relief, and Economic Security Act of 2020, except that repayments of any such loan may be (1) re-

amortized to reflect changes in payroll periods and/or (2) continued after the Participant incurs a Termination of Employment.

ARTICLE XI

SPECIAL PROVISIONS FOR TOP-HEAVY PLANS

11.01 <u>Determination of Top-Heavy Status</u>. The Plan shall be considered top-heavy for the Plan Year, if, as of the Determination Date:

(a) the Plan is not part of an Aggregation Group and the Key Employee Ratio, determined by substituting the "Plan" for the "Aggregation Group" each place it appears in Section 2.34, exceeds 60%, or

(b) the Plan is part of an Aggregation Group and the Key Employee Ratio of such Aggregation Group exceeds 60%;

11.02 <u>Minimum Contributions</u>. For any Plan Year in which the Plan is determined to be top-heavy within the meaning of Section 11.01, the Plan shall provide a minimum Employer contribution (consisting of Matching Contributions, nonelective Employer contributions, or both) for each Participant who is a Non-Key Employee and has not incurred a Severance from Employment by the end of the Plan Year in an amount equal to 5% of the Participant's Testing Compensation.

11.03 <u>Minimum Vesting</u>. For any Plan Year in which the Plan is defined to be top-heavy within the meaning of Section 11.01, each Participant during such Plan Year shall become 100% vested in all of his Accounts and shall remain fully vested in such Accounts after the Plan ceases to be top-heavy.

ARTICLE XII

PLAN ADMINISTRATION

12.01 Fiduciary Responsibility.

(a) The Benefit Strategy Committee shall be the Plan fiduciary with respect to the administration, and investment of the assets, of the Plan, except to the extent provided herein or that such authority has been delegated to an individual or committee.

(b) The Plan shall be administered by the Plan Manager, which shall be the Plan's "named fiduciary" and "administrator," as those terms are defined by ERISA, and its agent designated to receive service of process. All matters relating to the administration of the Plan, including the duties imposed upon the plan administrator by law, except those duties allocated to the Administrative Committee, and those duties relating to the control or management of Plan assets, shall be the responsibility of the Plan Manager. The Plan Manager or the Administrative Committee (to the extent of the duties of each under the Plan), as the case may be, shall have the power to interpret and construe the provisions of the Plan, and to decide such questions as may arise in connection with the operation of the Plan, including interpretation of ambiguous Plan provisions, determination of disputed facts, and application of Plan provisions to unanticipated circumstances. The determination of the Plan Manager or the Administrative Committee (to the extent of the duties of each under the Plan), as the case may be, shall be subject to review only for abuse of discretion.

(c) The Administrative Committee shall be responsible for reviewing and deciding appeals under the Plan, in accordance with Section 12.11(b) of the Plan.

(d) The Plan Manager shall be responsible for the day-to-day administration of the Plan and shall have the authority to adopt such rules, guidelines, forms and procedures, not inconsistent with the terms of the Plan, as deemed necessary and/or appropriate to the operation and/or administration of the Plan. The Plan Manager shall also be responsible for the reporting and disclosure requirements applicable to the Plan under ERISA, the Code and/or any other Federal, state or local law.

(e) The Investment Committee shall be responsible for all matters relating to the control and management of Plan assets, to the extent not assigned to the Trustee in the Trust Agreement or other instrument. The duties and responsibilities of the Investment Committee shall include, but shall not be limited to, the selection of the Investment Manager, and the monitoring of the performance of the Investment Manager and Trustee. The Investment Committee shall be a "named fiduciary" as that term is defined by ERISA.

(f) Notwithstanding any provision in the Plan to the contrary, effective October 23, 2020, Newport Trust Company shall be a Plan fiduciary within the meaning of section 3(21) of ERISA and an investment manager within the meaning of section 3(38) of ERISA with respect to the Unisys Common Stock Fund of the Plan, with the fiduciary authority and responsibility to:

(1) suspend or prohibit new investment of Participant or Employer contributions in the Unisys Common Stock Fund;

(2) suspend or prohibit the transfer of Participant Account balances into the Unisys Common Stock Fund;

(3) in connection with a determination that holding Unisys Stock is no longer prudent under ERISA, liquidate the Unisys Stock in the Unisys Common Stock Fund;

(4) direct that the proceeds from any liquidation of Unisys Stock pursuant to this subsection be invested on a temporary basis in the default investment option otherwise provided under the Plan (or in the absence of such a default option, in such investment option then available under the Plan as directed by the Company), pending Participant directions to the Trustee of the Plan with respect to the investment of such proceeds;

(5) suspend or prohibit the transfer of Participant Account balances out of the Unisys Common Stock Fund during any period in which Newport Trust Company is directing the liquidation of the Unisys Common Stock Fund;

(6) ensure compliance with all laws relating to the Plan's investment in Unisys Stock, including compliance with applicable securities laws and ERISA;

(7) in accordance with the terms of the Plan, periodically determine the portion of the Unisys Common Stock Fund to be maintained in cash or other liquid investments;

(8) to instruct the Trustee of the Plans with respect to the foregoing matters; and

(9) periodically report to the Investment Committee, either in writing or orally (as agreed to with the Investment Committee) regarding the actions taken by Newport Trust Company.

This subsection shall remain in effect until the agreement with Newport Trust Company is terminated by Newport Trust Company or the Company.

12.02 Appointment and Removal of Plan Manager and Committees.

(a) The Benefits Strategy Committee shall be a committee, consisting of not fewer than three persons appointed by the Chief Executive Officer of the Company. The Chief Executive Officer may, at any time, remove any member of the Benefits Strategy Committee, by written notice to such person, which notice shall be effective upon delivery, and appoint a successor. Persons serving on the Benefits Strategy Committee may, but need not, be employees of the Employer. Any Benefits Strategy Committee member may resign by giving written notice to the Chief Executive Officer of the Company, which notice shall be effective 30 days after delivery. The Chief Executive Officer of the Company shall promptly select a successor following the resignation or removal of the Plan Manager or of any Administrative Committee or Investment Committee member, if necessary to maintain both an Administrative Committee of at least one member.

(b) The Plan Manager, the Administrative Committee and the Investment Committee shall be individuals or committees designated as the plan manager, the administrative committee and investment committee, as applicable, under the Plan. The Benefits Strategy Committee may, at any time, remove the Plan Manager, any member of the Administrative Committee and/or the Investment Committee, by written notice to such person, which notice shall be effective upon delivery, and appoint a successor. The Plan Manager and persons serving on the Administrative Committee or the Investment Committee may, but need not, be employees of the Employer. The Plan Manager and any Administrative Committee or Investment Committee member may resign by giving written notice to the Benefits Strategy Committee, which notice shall be effective 30 days after delivery. The Benefits Strategy Committee shall promptly select a successor following the resignation or removal of the Plan Manager or of any Administrative Committee member, if necessary to maintain both an Administrative Committee and Investment Committee member.

12.03 <u>Compensation and Expenses of Plan Manager and Committees</u>. The Plan Manager and members of the Benefits Strategy Committee, the Administrative Committee or the Investment Committee who are Employees shall serve without compensation. The Plan Manager and members of the Benefits Strategy Committee, the Administrative Committee who are not Employees may be paid reasonable compensation for services rendered to the Plan. Such compensation, if any, and all ordinary and necessary expenses of the Plan Manager, the Benefits Strategy Committee, the Administrative Committee or the Investment Committee shall be paid from the Fund, unless paid by the Employer.

12.04 <u>Plan Manager and Committee Procedures</u>. The Plan Manager, the Benefits Strategy Committee, the Administrative Committee and the Investment Committee may enact such rules and regulations for the conduct of their business and for the administration of the Plan, as each may deem desirable. The Administrative Committee, the Benefits Strategy Committee, and the Investment Committee may act either at meetings at which a majority of its members are present, or by a writing signed by a majority of its members without the holding of a meeting. Records shall be kept of

the meetings and actions of the Administrative Committee, the Benefits Strategy Committee and the Investment Committee, and of the actions of the Plan Manager. Neither the Plan Manager, nor any Benefits Strategy Committee, Administrative Committee or Investment Committee member who is a Participant in the Plan shall vote upon, or take an active role in resolving, any question affecting only his benefits.

12.05 <u>Indemnification of the Plan Manager, the Benefits Strategy Committee, the Administrative Committee</u> and Investment Committee. The Plan Manager and each member of the Benefits Strategy Committee, the Administrative Committee and the Investment Committee shall be indemnified by the Company against costs, expenses and liabilities (other than amounts paid in settlement to which the Company does not consent) reasonably incurred by him in connection with any action to which he may be a party by reason of his service as Plán Manager or a member of the Benefits Strategy Committee, Administrative Committee or Investment Committee, except in relation to matters as to which he shall be adjudged in such action to be personally guilty of willful misconduct in the performance of his duties. The foregoing right to indemnification shall be in addition to such other rights as the Plan Manager or member of the Benefits Strategy Committee, Administrative Committee or Investment Committee may enjoy as a matter of law or by reason of insurance coverage of any kind, but shall not extend to costs, expenses and/or liabilities otherwise covered by insurance or that would be so covered by any insurance then in force if such insurance contained a waiver of subrogation. Rights granted hereunder shall be in addition to and not in lieu of any rights to indemnification to which the Plan Manager or member of the Benefits Strategy Committee, Administrative Committee or Investment Committee may be entitled pursuant to the bylaws of the Company. Service as Plan Manager or as a member of the Benefits Strategy Committee, Administrative Committee or Investment Committee shall be deemed in partial fulfillment of the member's function as an employee, officer or director of the Employer, if he serves in that capacity as well as in the role of Plan Manager or a member of the Benefits Strategy Committee, Administrative Committee or Investment Committee.

12.06 <u>Exclusive Benefit Rule</u>. The Plan Manager, the Benefits Strategy Committee and the Administrative Committee and the Investment Committee shall administer the Plan for the exclusive purpose of (a) providing benefits to Participants and their Beneficiaries and (b) defraying reasonable expenses of administering the Plan.

12.07 <u>Consultants</u>. The Plan Manager, the Administrative Committee and the Investment Committee may, and to the extent required for the preparation of reports shall, employ accountants, actuaries, attorneys and other consultants or advisors. The fees charged by such accountants, actuaries, attorneys and other consultants or advisors shall represent reasonable compensation for services rendered and shall be paid from the Fund unless paid by the Employer.

12.08 Payment of Plan Expenses. The expenses incurred by the Employer in connection with the operation of the Plan, including, but not limited to, expenses incurred by reason of the engagement of professional assistants and consultants, shall be expenses of the Plan and shall be payable by the Plan at the direction of the Plan Manager. The Employer shall have the option, but not the obligation, to pay any such expenses, in whole or in part, and, by so doing, to relieve the Plan from the obligation of bearing such expenses. Payment of any such expenses by the Employer on one occasion shall not bind the Employer to pay any similar expenses on any subsequent occasion. For the purpose of administrative convenience, the Employer may pay

certain expenses otherwise payable by the Plan, for which it shall seek reimbursement by the Trustee from the assets held in the Fund.

12.09 <u>Method of Handling Plan Funds</u>. All payments to the Fund shall be made by the employee of the Employer charged with that responsibility by the Investment Committee. All payments from the Fund shall be made by the Trustee.

12.10 <u>Delegation and Allocation of Responsibility</u>. To the extent permitted under the terms of the Trust Agreement or applicable law, the Trustee and any named fiduciary of the Plan may, by unanimous action in writing, delegate or assign any of its responsibilities for administering the Plan to one or more individuals or entities. In the event of any such delegation or allocation, the Trustee or any named fiduciary, as applicable, shall establish procedures for the thorough and frequent review of the performance of such duties. Persons to whom responsibilities have been delegated may not delegate to others any discretionary authority or discretionary control with respect to the management or administration of the Plan.

12.11 Claims Procedures.

(a) Initial Claim. In the event of a claim by a Participant or his or her Beneficiary with respect to the Plan, such claimant (himself or through his authorized representative) shall present his or her claim in writing to the Administrative Committee or its designee. The Administrative Committee or its designee shall, within 90 days after receipt of such written claim, make a determination and send a written or electronic notification to the claimant as to its disposition. If the Administrative Committee or its designee determines that special circumstances require an extension of time for processing the claim, the Administrative Committee or its designee shall be allowed an extension of time not to exceed 90 days from the end of the initial period and shall so notify the claimant in writing prior to the termination of the initial 90-day period, and shall indicate the special circumstances requiring an extension of time and the date by which to expect the benefit determination. In the event the claim is wholly or partially denied, such notification shall:

- (1) state the specific reason or reasons for the denial;
- (2) make reference to the specific provisions of the Plan upon which the denial is based;

(3) provide a description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why such material or information is necessary;

(4) set forth the procedure by which the claimant may appeal the denial of his or her claim and the applicable time limitations; and

(5) a statement of the claimant's rights to bring a civil action under section 502(a) of ERISA following an adverse benefit determination on appeal.

(b) <u>Review of Denial</u>. In the event a claimant wishes to appeal the denial of his claim, the claimant (or his or her authorized representative) may request a review of such denial by making application in writing to the Administrative Committee within 60 days after receipt of such denial, subject to any applicable extensions granted under guidance issued by the Department of Labor due to the national emergency attributable to the COVID-19 outbreak. Such review will take into account all comments,

documents, records, and other information submitted by the claimant relating to the claim, without regard to whether such information was submitted or considered in the initial benefit determination. Such claimant (or his or her duly authorized representative) may, upon written request to the Administrative Committee and free of charge, have reasonable access to, and copies of, all documents, records, and other information relevant to the claim for benefits. In addition, the claimant or his authorized representative may submit to the Administrative Committee written comments, documents, records and other information related to the claim for benefits. Appeals not timely filed shall be barred. Within 60 days after receipt of a written appeal, the Administrative Committee shall make a determination and notify the claimant of its final decision. If the Administrative Committee determines that special circumstances require an extension of time for processing the claim, the Administrative Committee shall be allowed an extension of time of up to an additional 60 days and shall so notify the claimant in writing (prior to the end of the initial period) the reason or reasons for such extension and the date by which a decision is expected. The final decision on review shall contain:

- (1) specific reasons therefor;
- (2) reference to the specific Plan provisions upon which it is based;

(3) a description of the claimant's right to receive, upon written request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to the claim for benefits;

- (4) a description of any voluntary appeals procedures offered by the Plan; and
- (5) a statement of the claimant's rights to bring a civil action under section 502(a) of ERISA.

If the Administrative Committee has not exceeded the time limitations set forth in this Section 12.11, the decision shall be final and conclusive on all persons claiming benefits under the Plan, subject to applicable law. If the claimant challenges the decision of the Administrative Committee, a review by a court of law shall be limited to the facts, evidence, and issues presented during the claims and appeals procedure set forth above. The claims and appeals process described herein must be exhausted before the claimant can pursue the claim in federal court. Facts and evidence that become known to the claimant after having exhausted the review procedure may be submitted for reconsideration of the review decision in accordance with the time limits established above. Issues not raised during the review process shall be deemed waived.

(c) Exhaustion of Claims Procedures and Time Period for Filing a Claim.

(1) <u>Deadline to File a Claim Against the Plan</u>. To be considered timely under the Plan's claims procedure, a claim must be filed with the Administrative Committee or its designee within one year after the claimant knew or reasonably should have known of the principal facts upon which the claim is based, subject to any applicable extensions granted under guidance issued by the Department of Labor due to the national emergency attributable to the COVID-19 outbreak. Any claim filed after the end of the period described in the preceding paragraph shall be time-barred.

(2) <u>Deadline to File Legal Action</u>. A claim or action (1) to recover benefits allegedly due under the provisions of the Plan or by reason of any law

(including, without limitation, a civil action under Section 502(a) of ERISA), (2) to enforce rights under the Plan, (3) to clarify rights to future benefits under the Plan, or (4) any other claim or action that relates to the Plan and seeks a remedy, ruling, or judgment of any kind against the Plan or a Plan fiduciary or party in interest may not be filed in any court until the claimant has exhausted the Plan's claim and appeal process for any and all reasons the claimant believes his claim should be approved. In addition, the deadline to file legal action is the later of (i) 24 months after the claimant knew or reasonably should have known of the principal facts on which the claim is based, or (2) six months after the claimant has exhausted the claims and review procedure. Any claim filed after the deadline described in the preceding sentence shall be time-barred.

12.12 <u>Revenue Credit</u>. If a Revenue Credit is payable to the Plan, the Trustee shall pay such amount to a Revenue Credit Account on a quarterly basis to be used in the manner specified in this Section.

(a) <u>Application of Account</u>. The Plan Manager may direct the Trustee to use amounts held in the Revenue Credit Account to reimburse the Company for expenses described in Section 12.08, or to pay such vendors, including the Trustee or third parties, directly in accordance with this Section and the terms set forth in the Trust Agreement. Amounts not used for such Plan expenses may be allocated to Participant Accounts in accordance with this Section, provided that such allocation shall not occur more frequently than quarterly.

(1) <u>Payment to Third Parties</u>. Upon receipt of payment instructions in good order from the Plan Manager, the Trustee shall redeem shares or units of investment options held in the Revenue Credit Account necessary to make such payments, and shall issue payment as soon as administratively feasible thereafter. The Revenue Credit Account shall not be used to offset, reimburse or pay: (i) expenses that have been deducted from Participant Accounts; or (ii) expenses that are accrued in the net asset value or mil rate of an Investment Fund.

(2) Allocation to Participants.

(A) Provided that the balance in the Revenue Credit Account amount exceeds \$1 per Participant on average, the Plan Manager may direct the Trustee, no more frequently than once per calendar quarter, to allocate balances to Participant Accounts; provided, however, that with respect to the last quarter of the Plan Year, the full remaining balance in the Revenue Credit Account (after the payment of Plan expenses and reimbursement of the Company for the payment of Plan expenses) shall be allocated to Participant accounts, effective as of the last day of the Plan Year, without regard to any minimum required balance.

(B) To the extent that the Plan Manager directs that balances in the Revenue Credit Account be allocated to Participants, the Trustee shall, in accordance with directions provided in the Trust Agreement, allocate to Eligible Participant Accounts a participant revenue credit ("Participant Revenue Credit") as soon as administratively feasible. Allocations shall be made pro rata based on Eligible Participant Account balances, exclusive of outstanding loan balances, as of the last day of the calendar quarter or Plan Year as designated by the Plan Manager (the "Crediting Date"). For purposes of Participant Revenue Credit allocations only, "Eligible Participant" means any Participant or Beneficiary (including an alternate payee to the extent provided under a Qualified Domestic Relations Order) with an Account balance greater than zero (prior to such Participant Revenue Credit allocation) on the business day immediately preceding the Crediting Date.

(b) <u>Investment</u>. The Revenue Credit. Account shall be invested in the fund designated in the Trust Agreement; provided, however, that, in the case of an allocation to Participant Accounts pursuant to subsection 12.12(a)(2) of amounts held in the Revenue Credit Account, such amounts shall be invested as set forth in the Trust Agreement.

(c) <u>Directions</u>. The Plan Manager shall provide direction to the Trustee when the Plan Manager wishes to use amounts held in the Revenue Credit Account for the payment of Plan expenses or allocation to Participants in the manner determined by the Trustee.

(d) <u>Definitions</u>.

(1) "Revenue Credit" means the amount determined in accordance with the Trust Agreement by which the fee offsets specified in the Trust Agreement exceed the recordkeeping fees described in such Trust Agreement.

(2) "Revenue Credit Account" means the suspense account under the Plan to which is deposited Revenue Credits.

12.13 Deemed Acceptance of Acts by Plan Fiduciaries. If the Plan Manager or other Plan fiduciary (as determined under ERISA) or an individual or entity with authority delegated by a Plan fiduciary, acts or fails to act with respect to a Participant, Beneficiary, Spouse or alternate payee or a Participant's Account under the Plan and the Participant, Beneficiary, Spouse or alternate payee has direct or indirect knowledge of such act or failure to act, the failure of the Participant, Beneficiary, Spouse or alternate payee to notify the Plan fiduciary (or the Plan fiduciary's delegate) within a reasonable period of time (but in no event more than 180 days) that such act or failure to act was incorrect or inconsistent with the intent or election of the Participant, Beneficiary, Spouse or alternate plan fiduciary's (or the Plan fiduciary's delegate) act or failure to act act was incorrect or be an acceptance and ratification of the Plan fiduciary's (or the Plan fiduciary's delegate) act or failure to act.

ARTICLE XIII

AMENDMENT AND TERMINATION

13.01 <u>Amendment</u>. The Plan may be amended at any time and from time to time by or pursuant to formal written action of (a) the Board, (b) the Compensation Committee of the Board, (c) the Chief Executive Officer of the Company, or (d) the Senior Vice President and Chief Human Resources Officer of the Company, the Senior Vice President and Chief Financial Officer of the Company, and the Senior Vice President and General Counsel of the Company acting as a committee; provided, however that amendments or modifications the annual cost of which exceed \$1,000,000 may not be made by the Senior Vice President and Chief Human Resources Officer of the Company, the Senior Vice President and Chief Financial Officer of the Company, and the Senior Vice President and General Counsel of the Company, the Senior Vice President and Chief Financial Officer of the Company, and the Senior Vice President and General Counsel of the Company, and amendments or modifications the annual cost of which exceed \$25,000,000 may not be made by the Chief Executive Officer of the Company, and amendments or modifications the annual cost of which exceed \$25,000,000 may not be made by the Chief Executive Officer of the Company, and amendments or modifications the annual cost of which will exceed \$50,000,000 can be made only by action of the Board. To the extent that the Corporate Delegation of Authority Chart or other action of the Board modifies the amendatory authority described in the preceding sentence, the Plan shall be deemed to have been amended in accordance with the Delegation of Authority Chart or such Board action.

Notwithstanding the foregoing, however, no amendment shall be effective to the extent that it has the effect of decreasing the Participant's Accrued Benefit or eliminating an optional form of benefit payment for benefits attributable to service before the later of the date the amendment is adopted or the date it becomes effective, except to the extent permissible under section 411(d)(6) of the Code and the regulations thereunder. If the vesting schedule of the Plan is amended, the nonforfeitable interest of a Participant in his Accrued Benefit, determined as of the later of the date the amendment is adopted or the date it becomes effective, shall not be less than the Participant's nonforfeitable interest in his Accrued Benefit determined without regard to such amendment. If the Plan's vesting schedule is amended, each Participant with three or more Years of Service may elect to have the nonforfeitable percentage of his Accrued Benefit computed under the Plan without regard to such amendment. The Participant's election shall be made within 60 days after the latest of (a) the date the amendment is adopted, (b) the date the amendment becomes effective, or (c) the date the Participant is given written notice of the amendment by the Company or the Trustee.

13.02 <u>Termination or Partial Termination</u>.

(a) <u>Right to Terminate Reserved</u>. While the Company intends to continue the Plan indefinitely, it reserves the right to terminate the Plan at any time by formal written action of (1) the Board, or (2) the Chief Executive Officer of the Company (or an entity or individual to whom such responsibility has been delegated). Further, any Employer may, at any time for any reason, withdraw from participation in the Plan, in whole or in part, by action of its governing board (or action of the committee or officer(s) to whom such authority has been delegated).

(b) <u>Treatment of Participants Upon Termination</u>. If the Plan is terminated or partially terminated, Accrued Benefits of the Participants affected thereby shall immediately vest and be nonforfeitable, to the extent funded. No employees of such Employer who are not then Participants may thereafter be admitted to the Plan, and the Employer shall make no further contributions to the Fund.

(c) <u>Liability of Employer</u>. The Employer shall have no liability in respect of payment under the Plan, except to pay over to the Trustee the contributions otherwise required under the Plan, and each Participant, his Beneficiary or alternate payee shall look solely to the Trust for distribution of benefits under the Plan.

(d) <u>Successor Employers</u>. Unless this Plan is terminated earlier, a successor employer of the Employees of the Employer may continue this Plan and Trust by joining with the Trustee in executing an appropriate supplemental agreement. Such successor employer shall ipso facto succeed to all the rights, powers, and duties of the Employer hereunder. In such event, the Plan shall not be deemed to have terminated and the employment of any Employee who is continued in the employ of such successor Employer shall be deemed not to have been terminated or severed for any purposes hereunder.

ARTICLE XIV

MISCELLANEOUS

14.01 <u>Merger, Consolidation or Transfer of Assets or Liabilities</u>. The Company reserves the right to merge or consolidate the Plan with any other defined contribution plan qualified under section 401(a) of the Code, or to transfer Plan assets or liabilities to any other qualified defined contribution plan, provided that the amount standing to the

credit of each Participant's, Beneficiary's and alternate payee's Accounts immediately after any such merger, consolidation or transfer of assets or liabilities shall be at least equal to the amount standing to the credit of the Participant's, Beneficiary's and alternate payee's Accounts immediately before such merger, consolidation or transfer, determined as if the Plan had then terminated. The Plan may be merged or consolidated with, or Plan assets and liabilities may be transferred to, another qualified defined contribution plan pursuant to formal written action taken by (1) the Board, or (2) the Chief Executive Officer of the Company (or an entity or individual to whom such responsibility has been delegated).

14.02 <u>Limited Purpose of Plan</u>. The establishment or existence of the Plan shall not confer upon any Employee the right to be continued as an Employee. The Employer expressly reserves the right to discharge any Employee whenever in its judgment its best interests so require.

14.03 <u>Nonalienation</u>. No benefit payable under the Plan shall be subject in any manner to anticipation, assignment, or voluntary or involuntary alienation. This Section 14.03 shall not preclude the Trustee from complying with the terms of (a) a Qualified Domestic Relations Order, (b) a federal tax levy made pursuant to section 6331 of the Code, (c) subject to section 401(a)(13) of the Code, a judgment relating to the Participant's conviction of a crime involving the Plan, or (d) subject to section 401(a)(13) of the Code, a judgment, order, decree, or settlement agreement between the Participant and the United States Department of Labor relating to a violation (or an alleged violation) of part 4 subtitle B of Title I of ERISA.

14.04 <u>General Distribution Requirements</u>. All distributions under the Plan shall be determined and made in accordance with the minimum distribution incidental death benefit requirements of the regulations under section 401(a)(9) of the Code. Notwithstanding any provision in the Plan to the contrary, all distributions shall be determined and made in accordance with section 401(a)(9) of the Code, including the incidental death benefit rules of section 401(a)(9)(G) of the Code, and final regulations issued thereunder, including Treas. Reg. §1.401(a)(9)-2 through 1.401(a)(9)-9.

14.05 <u>Facility of Payment</u>. If the Plan Manager, in his sole discretion, deems a Participant, Beneficiary or alternate payee who is entitled to receive any payment hereunder to be incompetent to receive the same by reason of age, illness, infirmity or incapacity of any kind, the Plan Manager may direct the Trustee to apply such payment directly for the benefit of such person, or to make payment to any person selected by the Plan Manager to disburse the same for the benefit of the Participant, Beneficiary or alternate payee. Payments made pursuant to this Section shall operate as a discharge, to the extent thereof, of all liabilities of the Employer, the Trustee, the Benefits Strategy Committee, the Administrative Committee, the Plan Manager, the Investment Committee and the Fund to the person for whose benefit the payments are made.

14.06 <u>Impossibility of Diversion</u>. All Plan assets shall be held as part of the Fund until paid to satisfy allowable Plan expenses or to provide benefits to Participants, their Beneficiaries or alternate payees. It shall be impossible, unless Section 4.10, 14.07 or 14.10 applies, for any part of the fund to be used for, or diverted to, purposes other than the exclusive benefit of the Participants, their Beneficiaries or alternate payees or the payment of the reasonable expenses of the administration of the Plan or of the Fund or both, and the Fund shall continue for such time as may be necessary to accomplish the purposes for which it was established.

14.07 <u>Unclaimed Benefits</u>. If a Participant or Beneficiary to whom a benefit is payable under the Plan cannot be located following a reasonable effort to do so by the Trustee, such benefit shall be forfeited but shall be reinstated if a claim therefor is filed by the Participant, Beneficiary or alternate payee.

Mistaken Payment. If it is determined that a Participant, Beneficiary, Spouse or alternate payee has 14.08 received the incorrect payment(s) for any reason, overpayments shall be charged against, and underpayments shall be added to, any benefits otherwise payable to any such individual. The Plan has a right of reimbursement against any person who receives or holds a payment from the Plan in excess of the amount to which a Participant, Beneficiary, Spouse or alternate payee is entitled under the terms of the Plan. Any such person has an affirmative obligation to repay such excess amounts to the Plan, and holds such amounts, legally or constructively, in constructive trust and pursuant to an equitable lien by agreement for the benefit of the Plan, regardless of such person's current possession, or the absence thereof, of such amounts. In addition, the Plan may recover the amount overpaid (plus interest) in any manner determined by the Plan Manager to be in the best interests of the Plan, including, but not limited to, legal action against the recipient and/or holder of the overpayment or by offset of up to 100% of other or future benefits payable to or with respect to the Participant, Beneficiary, Spouse or alternate payee under the Plan. Any fault on the part of such person or the Plan Manager or other agent or representative of the Plan shall be neither a condition to nor a defense against the Plan's right to recover such excess amounts. Receipt of such excess amounts by any such person shall constitute that person's waiver of reliance or estoppel as a defense against the Plan's rights to recover such excess amounts. The provisions of this Section shall apply to all past or future overpayments.

14.09 <u>Controlling Law</u>. The Plan shall be construed and enforced in accordance with the laws of the Commonwealth of Pennsylvania, without regard to any choice of law provisions, to the extent not preempted by federal law, which shall otherwise control.

IN WITNESS WHEREOF, and as evidence of the adoption of the Plan as amended and restated herein, Unisys Corporation has caused this instrument to be executed by its duly authorized representatives.

By:	/s/ Katie Ebrahimi	 Dated: December 16, 2022
,	Katie Ebrahimi	
	Chief Human Resources Officer	

<u>/s/ Claudius Sokenu</u> Dated: December 16, 2022 Claudius Sokenu Chief Administrative Officer, General Counsel, and Corporate Secretary

<u>/s/ Deb McCann</u> Deb McCann Chief Financial Officer

Exhibit 31.1

CERTIFICATION

I, Peter A. Altabef, certify that:

1. I have reviewed this quarterly report on Form 10-Q of Unisys Corporation;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 2, 2023

Name: Title: /s/ Peter A. Altabef Peter A. Altabef Chair and Chief Executive Officer

Exhibit 31.2

CERTIFICATION

I, Debra McCann, certify that:

1. I have reviewed this quarterly report on Form 10-Q of Unisys Corporation;

2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 2, 2023

Name: Title: /s/ Debra McCann Debra McCann Executive Vice President and Chief Financial Officer

Exhibit 32.1

CERTIFICATION OF PERIODIC REPORT

I, Peter A. Altabef, Chair and Chief Executive Officer of Unisys Corporation (the "Company"), certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that:

(1) the Quarterly Report on Form 10-Q of the Company for the quarterly period ended March 31, 2023 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m); and

(2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 2, 2023

/s/ Peter A. Altabef Peter A. Altabef Chair and Chief Executive Officer

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

Exhibit 32.2

CERTIFICATION OF PERIODIC REPORT

I, Debra McCann, Executive Vice President and Chief Financial Officer of Unisys Corporation (the "Company"), certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that:

(1) the Quarterly Report on Form 10-Q of the Company for the quarterly period ended March 31, 2023 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m); and

(2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 2, 2023

/s/ Debra McCann Debra McCann Executive Vice President and Chief Financial Officer

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.